

Temi di discussione

(Working Papers)

Firms' inflation expectations and investment plans

by Adriana Grasso and Tiziano Ropele





Temi di discussione

(Working Papers)

Firms' inflation expectations and investment plans

by Adriana Grasso and Tiziano Ropele

Number 1203 - December 2018

The papers published in the Temi di discussione series describe preliminary results and are made available to the public to encourage discussion and elicit comments.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Federico Cingano, Marianna Riggi, Emanuele Ciani, Nicola Curci, Davide Delle Monache, Francesco Franceschi, Andrea Linarello, Juho Taneli Makinen, Luca Metelli, Valentina Michelangeli, Mario Pietrunti, Lucia Paola Maria Rizzica, Massimiliano Stacchini. *Editorial Assistants:* Alessandra Giammarco, Roberto Marano.

ISSN 1594-7939 (print)

ISSN 2281-3950 (online)

Printed by the Printing and Publishing Division of the Bank of Italy

FIRMS' INFLATION EXPECTATIONS AND INVESTMENT PLANS

by Adriana Grasso^{*} and Tiziano Ropele^{**}

Abstract

In past years there have been suggestions for monetary policy to engineer higher inflation expectations to stimulate spending. We examine the relationship between the inflation expectations of firms and their investment plans using Italian business survey data over the period 2012-2016. We show that higher expected inflation is positively correlated with firms' willingness to invest. In our baseline specification, a one percentage point rise in expected inflation is associated with a higher probability of reporting higher investment plans by 4.0 percentage points. This expansionary effect operates through the standard interest rate channel and its magnitude is positively correlated with firms' liquidity and debt position.

JEL Classification: E22, E31, E58.

Keywords: investment expenditure, inflation expectations, survey data.

Contents

1. Introduction	5
2. Description of data sources	8
2.1 The SIGE	9
2.2 The Italian Credit Register	12
2.3 The Company Accounts Data system	12
3. Basic descriptive statistics	13
4. Econometric strategy	14
5. Baseline estimation results	17
6. Robustness checks	19
7. Inspecting the channels	21
7.1 Results	24
8. Conclusions	
References	
Figures and tables	
Online appendix	45

^{*} Bank of Italy, Economic Outlook and Monetary Policy Directorate; LUISS University.

^{**} Bank of Italy, Economic Research Unit, Milan Branch.

"Inflation means distribution of buying power, credit expansion, rising prices, restoration of markets, increased business turnover. But its benefits will not be distributed equally. Inflation will mean most to the business man who meets it half way, who increases his business pace to keep up with the accelerated dollar, who unleashes his jealously guarded cash reserves and credit [...]." Business Week, May 10, 1933

1 Introduction¹

It is widely acknowledged that persistently low inflation developments pose substantial macro-economic risks especially when the monetary policy is constrained by the effective lower bound (ELB) on nominal interest rates. The main danger is that if a low inflation outlook becomes entrenched in the private sector's expectations then firms and households will perceive a higher real interest rate and reduce spending. This in turn may give rise to a spiral of lower demand and lower prices, entangling the economy in a deflation trap. To avoid these risks and shore up the economic activity many economists and policymakers have suggested the adoption of policies to engineer higher inflation expectations.² Yet, beyond the interest rate channel, higher anticipated inflation may activate other mechanisms whose ultimate effects on agents' economic decisions are theoretically ambiguous. For example, higher expected inflation can depress spending because it acts as an implicit tax on liquid assets (*negative* wealth effect) and could portend future higher interest rates.³ Conversely, expansionary effects from higher inflation expectation may result from the additional borrowing capacity created by the erosion of nominal debt burden (*positive* wealth effect) or because of a Tobin effect whereby agents substitute away liquid assets with durable goods or other fixed assets.

In this paper, we provide new econometric evidence on both the sign and the magnitude of the relationship between inflation expectations of firms in Italy and their investment expenditure plans. For this purpose we use a novel dataset that we construct by merging firmlevel data coming from three distinct sources: the Italian Survey of Inflation and Growth

¹ Acknowledgments. We are grateful to seminar participants at LUISS University, University of Milano-Bicocca, Einaudi Institute for Economics and Finance and Bank of Italy and participants at the 7th Ifo Conference on "Macroeconomics and Survey Data". We wish to thank Davide Arnaudo, Rudi Bachmann, Pierpaolo Benigno, Robert Chirinko, Olivier Coibion, Jean-Paul L'Huillier, Francesco Lippi, Stefano Neri, Juan Passadore, Facundo Piguillem, Massimiliano Rigon, Paola Rossi, Alessandro Secchi, Eric Sims, Andrea Stella, Patrizio Tirelli and Michael Weber for helpful comments and suggestions. The views expressed here should not be interpreted as representing the views of the Bank of Italy.

² E.g. Krugman (1998), Eggertsson and Woodford (2003) and Eggertsson (2008).

³ Higher inflation expectations can also lead to higher macroeconomic uncertainty in response to which agents increase precautionary savings thus reducing spending.

Expectations (SIGE), the Italian Credit Register and the Company Accounts Data System. In particular, every quarter SIGE collects quantitative inflation expectations over various forecasting horizons and qualitative measures of the expected annual change in investment expenditure. The latter are gathered from the responses to a question that admits five ordered options from "much lower" to "much higher". SIGE also contains a rich set of information on idiosyncratic and aggregate evaluations that we control for in our empirical analysis to cope with the endogeneity problem of inflation expectations and thus ensure that the identifying variation in expected inflation is as much as possible not related to other factors that impact on firm investment plans. Estimating a battery of ordered probit regressions on pooled data over the period from 2012Q4 to 2016Q4 we find robust evidence indicating a highly significant and positive relationship between inflation expectation of firms and their investment expenditure plans. In the baseline estimates, a one percentage point increase in the 6-month ahead inflation expectations is associated with a higher predicted probability to report higher investment expenditure plans by 4.3 percentage points and by 3.8 percentage points when using the 12month ahead inflation expectations. With longer-term expectations the effects remain significant albeit the magnitude decreases somewhat.

Once established the existence of a significant and positive nexus between inflation expectations of firms and their investment attitude, we use the information from the Italian Credit Register, specifically the firm-level nominal interest rate on loans, and the information from the Company Accounts Data System, specifically an indicator of firm default risk (which we use as a proxy for the nominal borrowing cost), the cash ratio (sum of cash and marketable securities to current liabilities) and the debt ratio (current liabilities to total assets), to shed light on the channels through which inflation expectations may affect the investment plans of firms. In this regards, we find evidence supporting the real interest rate channel. The effect of the real interest rate on investment expenditure plans is significant and displays, as predicted by the theory, the negative sign. Similarly, including in the specification the nominal interest rate (or its proxy) and the inflation expectation as separate regressors we still obtain effects that are statistically significant and that confirm the expansionary effects of higher expected inflation. Furthermore, we find that the magnitude of the relationship between inflation expectations and investment plans vary positively with the levels of liquidity and debt of firms. A one percentage point increase in the 6-month ahead inflation expectations is associated with a higher predicted probability to report higher investment expenditure plans by 4.0 and 5.9 percentage points when the cash ratio is 5 and 60 percent, respectively. The estimated coefficient of expected inflation rises from 3.0 to 7.1 percent when the level of the debt ratio is 20 or 80 percent, respectively. Similar estimated coefficients are obtained when the estimation is carried out using the 12-month ahead inflation expectations.

Overall, these results bear important policy implications. The robust finding that higher inflationary expectations exert expansionary effects on business investment decisions lends support to the policy measures that operate primarily through the expectations channels as means to effectively stimulate the economy. Furthermore, the evidence according to which the magnitude of the effects of higher inflation expectations varies in relation with the liquidity and debt position of firms suggests a potential synergy between policies aimed at engineering higher inflation expectations (like for instance monetary or fiscal forward guidance announcements) and policies meant to facilitate the channeling funds to firms (like for instance the targeted longer-term refinancing operations launched by the European Central Bank in the past few years or any other measure meant to ease the condition of access to credit).

These policy implications must be taken with some caution for at least two reasons. First, we provide reduced-form estimates and thus our findings may be subject to the Lucas critique. Second, our empirical analysis covers a specific sample period, which includes part of the sovereign debt crisis, historically very low levels of the policy rates and the presence of extraordinary interventions of monetary policy. This is to say that our results may not hold in other phases of the business cycles or under other economic circumstances.

Our paper is related to the growing empirical literature that studies how inflation expectations of economic agents relate to their economic decisions. So far, much of this work has focused on households, in part due to the greater availability of household surveys of inflation expectations. Bachmann et al. (2015) using the Michigan Survey of Consumers find that the effects of higher inflation expectations on consumers' readiness to spend are generally small and statistically insignificant, and significantly negative when the economy is at the effective lower bound. Burke and Ozdagli (2013) find similar results using data from the New York Fed/ RAND-American Life Panel household expectations survey. Subsequent works have found instead positive correlations between expectations and consumption using the New York Fed's Survey of Consumer Expectations (Crump et al., 2015), a German survey of households

(Dräger and Nghiem, 2016; D'Acunto et al., 2018), a broader cross-section of European households (Duca et al., 2017) and a Japanese survey of households (Ichiue and Nishiguchi, 2015).

Fewer studies have examined the effects of inflation expectations of firms on economic decisions.⁴ Coibion, Gorodnichenko, Kumar and Pedemonte (2018) use an experimental design in a quantitative survey of firms in New Zealand to assess how exogenous variation in inflation expectations of managers from an information treatment affects their subsequent choices over prices, wages, employment and investment. Similarly, Coibion, Gorodnichenko and Ropele (2018) exploit a unique design feature of SIGE (consisting in the fact that since September 2012 a randomly chosen subset of firms has been repeatedly "treated" with information about recent inflation whereas other firms have been not) to study the causal effect of inflation expectations on firms' economic decisions and forecasts. Ropele (2018) is another recent study to use SIGE to examine the nexus between inflation expectations and the expected price-setting behavior of firms while Cloyne et al. (2016) conduct a similar investigation using survey data for UK manufacturing firms. Our empirical analysis complements these empirical studies by providing new evidence on the relationship between inflation expectations of firms and their investment expenditure plans, which represent a fundamental determinant of business cycle fluctuations, and on the channels through which such effects occur.

The paper is organized as follows. Section 2 provides information about the survey and other data sources. Section 3 presents basic descriptive statistics of the key variables to our analysis. Section 4 discusses the econometric strategy we employ to study the effects of firms' inflation expectations, while sections 5 and 6 present basic estimation results and robustness checks. Section 7 reports the estimation results regarding the transmission channels of inflation expectations. Finally, section 7 concludes.

2 Description of Data Sources

Firm-level data used in this paper come from three sources: (*i*) the SIGE, (*ii*) the Italian Credit Register and (*iii*) the Company Accounts Data System.

⁴ Other studies have analyzed the formation of inflation expectations of firms (e.g. Richards and Verstraete 2016 using Canadian survey data; Bryan et al. 2015 using data for firms in southeastern United States; Bartiloro et al. 2017 and Conflitti and Zizza 2018 survey data from SIGE) or the anchoring of inflation expectations (e.g. Kumar et al. 2015 using New Zeland survey data).

2.1 The SIGE

The SIGE is a quarterly business survey run since December 1999 by the Bank of Italy in collaboration with the financial newspaper *Il Sole 24 Ore*. The survey covers a sample of about 1,000 Italian firms with at least 50 employees, which is stratified by sector of economic activity (industrial, non-financial private services and construction⁵), geographical area (North-West, North-East, Centre, South & Islands) and number of employees (50-199, 200-999, 1000 and over). The list of firms used to extract the sample is drawn from the Bureau Van Dijk's Aida database and is updated on average every five years.

The survey is conducted by a specialist firm that distributes the questionnaire to company managers who are best informed about the topics covered in the survey. About 90 percent of the data is collected through computer assisted web interviews in the form of an online questionnaire featuring a purpose-designed interface, while the remaining 10 percent are collected through computer assisted telephone interviews. Data are collected in the first three weeks of March, June, September and December. The response rate is on average 45 percent.

The purpose of the survey is to obtain current or prospective information on firms' assessments of macroeconomic matters as well as various aspects of their business activity. Most of the data is qualitative and typically admits three or more possible answers (for example: worse, the same, better).

Investment expenditure expectations

Since December 2012 firms are asked the expected change in their investment expenditure. In the survey waves of March, June and September the question is formulated as follows:

What do you expect will be the nominal expenditure on (tangible and intangible) fixed investment in the current year compared with that of last year? \Box much lower; \Box a little lower; \Box about the same; \Box a little higher; \Box much higher.

In the survey wave of December it reads as:

What do you expect will be the nominal expenditure on (tangible and intangible) fixed investment next year compared with that in the current year? \Box much lower; \Box a little lower; \Box about the same; \Box a little higher; \Box much higher.

⁵ Construction firms have been included in the survey since March 2013.

Several remarks are in order. First, in both formulations firms are asked to indicate the expected change in fixed investment expenditure on an annual basis but over the current calendar year relative to the previous year in the former case and over the subsequent calendar year in the latter case. Hence, the *actual* forecasting horizon varies throughout the survey waves between 3- to 12month ahead. Second, firms answer this question by choosing among five ordered qualitative categories and a footnote added to the question (not reported above) invites the respondent to use the categories "much higher" and "much lower" if in any of the two periods investment expenditure is zero. Third, the question asks the expected change in investment expenditure in nominal terms. Needless to say, we are interested in studying the effects of inflation expectations of firms on their expected investment expenditure in real terms. Unfortunately, with the available information in SIGE we cannot directly address this problem. Yet, we argue that this concern should not represent an issue in this analysis for the following considerations. First, using the annual firm-level data from the Survey of Industrial and Service Firms, which is another business survey conducted by the Bank of Italy, over the period from 2012 to 2016 firms on average expected the price of investment goods in the next 12 months to grow in the range 1.1-2.2 percent. Furthermore, focusing on the subset of firms that predicted to reduce in the next 12 months the nominal investment expenditure, the expected change was on average equal to -45 per cent.⁶ These results indicate that the expected change in investment expenditure was essentially driven by actual purchase of investment goods rather than their price dynamics. A similar conclusion can be drawn from panel F of Figure 1 where it is shown that the actual growth rates of non-residential investment expenditure in Italy in nominal vis- \hat{a} -vis real terms track each other pretty closely.

Consumer price inflation expectations

Since the inception of SIGE firms have been asked to report a quantitative forecast of the Italian Harmonized Index of Consumer Price in terms of its 12-month ahead annual percentage change. Later on, the question on inflation expectations has been enriched to consider other

⁶ The time development of the first decile of the empirical distribution of the decline rate of firms' expected nominal investment expenditure is larger in absolute value than the expected growth rate in the price of investment goods.

forecasting horizons: 6-month ahead, 2-year ahead and 2-year ahead from 2 years. For example, in survey wave of December 2015 the question was formulated as follows:

In October consumer price inflation, measured by the 12-month change in the harmonized index of consumer prices was 0.3 per cent in Italy and 0.1 per cent in the euro area. What do you think it will be in Italy in: June 2016? __. _%; December 2016? __. _%; December 2017? __. _%; on average between December 2018 and December 2020? __. %.

The first thing to note here is that firms provide numerical point forecasts with one decimal digit of precision. As discussed in Coibion, Gorodnichenko, Kumar and Pedemonte (2018), this feature of the survey design is desirable as it allows respondents to freely choose their inflation expectation.⁷ At the same time, to channel firms' responses towards plausible figures the question provides a nominal anchor in the form of the latest official inflation rates for Italy and for the Euro Area. To ensure a uniform informational framework, interviews are started just after the announcement of the latest provisional inflation figure referred to the preceding month. Since September 2012 the question has been slightly modified to evaluate the effects of presenting firms the nominal anchor. The sample of firms has been randomly split in two groups with two thirds of firms receiving the information treatment and the remaining one third being instead uninformed. In the present study we only consider the responses by informed firms, which constitute the largest sub-group.

Other information from SIGE

As discussed earlier, SIGE asks firms many other questions that cover matters related to firm activity (e.g. the conditions of access to credit, the expected labor demand, the own-product expected demand or own-product expected price change) or the general economic situation in Italy. The list of all these questions that we later use in the empirical analysis is reported in Table 1. Furthermore, SIGE asks firms to report several structural characteristics regarding the number of employees, the sector of economic activity (industry, non-financial private services and construction), the geographical location of the firm (North-West, North-East, Centre, South & Islands) and the share of sales from exports (no export, export share in total sales is 1 to 33

⁷ In other business surveys firms are allowed to provide only a qualitative forecast or select their answer choosing from a set of quantitative ranges.

percent, export share is 34 to 66 percent, export share is 67 percent or more). Also these demographic characteristics will be used in the empirical investigation.

2.2 The Italian Credit Register

The second source of firm-level data is the subsection TAXIA of the Italian Central Credit Register, which contains detailed quarterly information on loans provided by a representative sample of financial intermediaries (about 200 Italian banks and 10 branches and subsidiaries of foreign banks).⁸ Using the individual data from TAXIA we compute the firm-level nominal borrowing cost (inclusive of fees and commissions) on new term loans obtained in the quarter. We focus on new term loans for two reasons. First, term loans represent the technical form most commonly used to finance investment projects. Second, the cost of new loans obtained in each quarter represents an accurate measure of the financing condition in that specific point in time and possibly a good proxy for the financing condition prevailing in the near future. That said, this data choice comes with the inconvenience of generating a large number of missing observations as in practice firms neither demand nor obtain loans in every period. To tackle this issue, in the empirical part of this paper we also provide estimation results obtained by proxying the nominal borrowing cost with a measure of firm default risk.

2.3 The Company Accounts Data System

The third source of firm-level data is the Company Accounts Data System (CADS), which is administered by Cerved Group and includes balance sheet information for all Italian limited liability companies. From CADS we obtain three annual indicators.

The first one is an indicator of the risk profile or default risk of each firm (which we refer to as the *score*). The *score* is computed annually by Cerved Group using discriminant analysis based on a series of balance sheet indicators (assets, rate of return, debts etc.) according to the methodology described in Altman (1968) and Altman et al. (1994) and takes integer values from 1 (when a firm is classified as "very sound") to 9 (when a firm is classified as "very high risk").⁹ As discussed in Panetta et al. (2009), although the *score* indicator becomes available with a delay of about 15 months it is nonetheless widely used by Italian banks to assess firm default risk and

⁸ Only firms whose total lending from a single bank exceeds 30,000 euro are reported in TAXIA.

⁹ The other values in the scale are: 2 = "sound", 3 = "above average solvency", 4 = "solvent", 5 = "vulnerable", 6 = "high vulnerability", 7 = "risky" and 8 = "high risk".

price loans. In light of this consideration and also to maintain the largest number of SIGE observations, in several econometric specifications we replace the quarterly nominal borrowing cost with the annual *score* (lagged one year).

Using the balance sheet data from CADS we also construct two ratios to gauge the shortterm liquidity position and the short-term debt position of firms. The short-term liquidity is measured as the ratio of the sum of cash and marketable securities to current liabilities, i.e. debts that are due to be paid within one year. This ratio is commonly known as the *cash ratio*. The short-run debt position is instead measured as the ratio of current liabilities to total assets. We will use these ratios in Section 7 to investigate to what extent the effects of inflation expectations on investment expenditure plans of firms depend on these balance sheet characteristics.

3 Basic descriptive statistics

In Table 2 we report basic descriptive statistics of selected variables computed by pooling the data from 2012Q4 to 2016Q4.¹⁰ As shown in row (1), over the sample period nearly half of firms reported not to expect changes in their investment expenditure while 40 percent of the respondents indicated little changes (24.5 and 15.5 percent of respondents expected a little higher or a little lower investment plans, respectively). Few firms responded choosing the boundary categories much lower and much higher. As illustrated in Figure 1 (panels A-E), in contrast to the little time variation in the share of firms expecting no change in their investment plans, which remained in the range 40-50 percent, the quota of those that indicated a little lower or higher investment progressively decreased in the former case (from about 25 to 10 percent) and rose in the latter (from about 20 to 30%). The frequency of firms that reported much lower investment expenditure stayed at about 15 percent until the second quarter of 2013 and halved thereafter while the share of firms that expected much higher investment represented was negligible (about 4 percent).¹¹

The gradual shift towards firms with more positive investment plans is also depicted in panel F of Figure 1, which reports the net percentage of firms expecting higher investment

¹⁰ As already discussed, in 2013Q3 the question on expected investment expenditure was not presented to firms. For this reason, all descriptive statistics are computed without taking into account this survey wave.

¹¹ Given the small number of responses falling in the boundary categories, we make the question on expected investment expenditure trichotomous by merging the categories "much lower" and "a little lower" in "lower" and similarly "much higher" and "a little higher" in "higher". We return to the original formulation in Section 6 when we conduct some robustness checks.

expenditure. In the same panel we also report the actual annual growth rates of nonresidential investment expenditure in Italy in real as well as nominal terms. It is worth noting that in either case the net percentage tracks fairly closely the observed investment dynamics.

Turning to firms' inflation expectations, as shown in rows 10 and 11 of Table 1, over the sample period firms expected the aggregate prices to increase by 0.84 percent in the next 12 months and 0.71 percent in the next 6 months. The degree of disagreement in inflation expectations among firms, measured by the standard deviation or the interquartile range, does not reveal substantial differences between the two forecasting horizons. In the top panels of Figure 2 we report the time evolution of the average and standard deviation of firms' inflation expectations. Until 2014Q4 firms expected prices in Italy to gradually decelerate and then virtually stabilize. Likewise, the degree of disagreement across firms declined in the first part of the sample period and then flattened out (except for the spike recorded in 2015Q1).

Once discussed separately the basic descriptive statistics for investment and inflation expectations, we now turn to provide a preliminary assessment of the relationship between these two variables. In the bottom panels of Figure 2 we compare the average inflation expectation for firms that expected to lower investment and with that of firms that reported higher expected investment. More clearly since 2013Q4, there appears a positive relationship between inflation and investment expectations suggesting that firms that expected higher inflation also planned higher investment expenditure. As reported in Appendix Table 1, over the sample period the average inflation expectation formulated by firms that expected to increase investment is about 10 basis points higher than for firms that planned to reduce it.¹²

4 **Econometric strategy**

In order to examine the relationship between of the inflation expectations of firms and the categorical variable that defines their investment expenditure plans we estimate a series of ordered probit regressions. We assume the existence of an unobserved continuous measure of firms' investment expectation (INVEST $_t^{i,*}$) for which we only observe discrete outcomes

¹² In Appendix Table 1 we show basic descriptive statistics to illustrate the correlation between the expected investment plans of firms and other firm variables. It turns out that on average firms that predict "higher" investment plans also indicate a more favorable idiosyncratic as well as macroeconomic outlook. Furthermore, they pay a lower nominal interest rate and expect a lower real interest rate compared to firms that predict to reduce investment. Finally, they relatively more liquid and less indebted.

represented by the response categories to the survey question on investment expenditure. In its simplest format, the latent regression is given by

$$INVEST_t^{i,*} = \beta INFL_t^{i,h} + \gamma \mathbf{X}_t^i + error_t^i$$
(1)

where $\text{INFL}_{t}^{i,h}$ represents the inflation expectation of firm *i* at horizon *h* formulated in period *t*, \mathbf{X}_{t}^{i} represents a vector of firm-specific and aggregate controls which we discuss in more detail below and β and γ are coefficients to be estimated.

Building our econometric analysis on a regression specification like (1) exposes us to two potential dangers. The first one regards the problem of reverse causality between investment expenditure plans and inflation expectations of firms. It is true that from a macroeconomic perspective investment and inflation expectations are determined simultaneously thus making hard to establish a direction of causality. That said, in the present context this danger should not represent a concern as here we relate the expectations of firms on a macroeconomic matter (i.e. the expected growth rate of the consumer price index in Italy) with their expectations on a microeconomic matter (i.e. the expected change of their investment expenditure). We believe highly unlikely that individually firms may think that their investment plans can materially affect the consumer price index. This consideration is reinforced by the fact that the Italian production structure is largely characterized by small and medium-sized firms.

The second danger regards the endogeneity of inflation expectations. If firms' inflation expectations are influenced by variables that are not included in the regression specification than the estimate of the effect of inflation expectations will be biased. Therefore, in order to avoid as much as possible this problem we need to control for determinants of investment that may be correlated with inflation expectations.¹³ These variables can be cross-sectional or aggregate in nature.

Fortunately, SIGE contains a rich set of information on expectations and evaluations of firms for which we can control in our regression specifications. In order to capture movements along or shifts of the "perceived" Phillips curve that can give rise to changes in inflation

¹³ A similar empirical strategy to address the endogeneity problem of inflation expectations, based on the inclusion of numerous control variables in the regression specification, is pursued in Bachmann et al. (2015). Another way to proceed would be to exploit a source of exogenous variation in the inflation expectations of agents. This route is for instance followed in Coibion, Gorodnichenko and Ropele (2018) and D'Acunto et al. (2018).

expectations of firms we consider the following set of control variables. We include the (qualitative) assessment of firms regarding the current macroeconomic outlook in Italy compared with previous 3 months ("worse", "the same", "better"). We include the (qualitative) assessment of firms regarding the direction and intensity that several factors exert on their expected price-setting behavior in the next 12 months. In particular we focus on two factors: the cost of labor and the price of raw materials. As reported in Table 1, firms respond by choosing among seven options, ranging from "downward, high" to "upward, high".

As documented in various studies, the expectations of agents may display systematic errors and be biased towards optimism or pessimism (e.g. Bachmann and Elstner, 2015). A positive correlation between investment expenditure plans and inflation expectations could be the result of firms being optimistic or pessimistic by nature. Not controlling for this attitude of firms would tend to induce a positive or a negative correlation between expected inflation and the error term. We address the "optimist/pessimist" problem by including in our vector of controls the (qualitative) assessment of firms about the current conditions to invest compared with previous 3 months ("worse", "the same", "better") and the evaluation of firms regarding the probability of an improvement of the macroeconomic outlook in Italy in the next 3 months ("zero", "1-25 percent", "26-50 percent", "51-75 percent", "76-99 percent" and "100 percent").

The control vector also needs to account for purely aggregate covariates (such as shocks, trends and even the provision of the nominal anchor to firms). Similarly to the logic discussed above, a strong economy may be positively correlated with investment expenditure plans but also with inflation expectations of firms. To this end, we also include time fixed effects.

Beyond the controls just described above, we exploit other information from SIGE that regard the idiosyncratic situation of firms. We include the evaluation on the current conditions of access to credit compared with previous 3 months ("worse", "unchanged", "better"), the evaluation on expected demand of labor in the next 3 months ("lower", "unchanged", "higher"), the evaluation on the expected demand in the next 3 months ("lower", "unchanged", "higher") and the expected price-setting behavior in the next 12 months (firms provide a quantitative forecast of the expected percentage change in selling price). Finally, we also include several demographic characteristics of firms (size, sector, area and openness to exports).

5 Baseline estimation results

In this section we present the baseline estimation results from ordered probit regressions as laid out in the previous section. For our baseline specifications we focus on the effects of inflation expectations of firms over a 6-month and 12-month ahead horizons. Unless otherwise specified, the marginal effects are evaluated at the sample mean values for the continuous regressors and at the sample modal categories for the qualitative regressors.

Table 3 shows the estimated coefficients as well as the marginal effects of inflation expectations. The marginal effects of the other control variables (except for the time fixed effects are shown in Appendix Tables 2 and 3).¹⁴ The marginal effects have the economic interpretation as the change in the predicted probability of reporting any of the three response categories ("lower", "about the same", "higher") for a one percentage point increase in expected inflation.

We find a positive and statistically significant coefficient on inflation expectations ($\beta = 0.116$ in the case of 12-month ahead expectations and $\beta = 0.131$ in the case of 6-month ahead expectations), suggesting that higher inflation expectations are associated with firms being more willing to raise their investment expenditure plans. These results are confirmed by the estimated marginal effects of inflation expectations on the predicted probability to report "lower" or "higher" investment expenditure plans. Specifically, a one percentage point increase in the 12-month ahead inflation expectations is associated with a lower (higher) predicted probability of reporting lower (higher) investment expenditure by about 3.3 (3.8) percentage points. The marginal effects are somewhat larger when using the 6-month ahead inflation expectations. Finally, higher expected inflation also appears negatively related (with a statistically significant at 5 percent) with the probability of reporting no change in investment expenditure. In this case, thought, the effects are quantitatively small.

In Section 3 we showed that in the sample period under consideration the cross-sectional mean of inflation expectations initially declined and then stabilized at very low levels. Though the range of variation in the mean expected inflation is not particularly wide, the evaluation of the marginal effects at the sample mean of inflation expectations could be too restrictive. We

¹⁴ As shown in Appendix Tables 2 and 3, the marginal effects of most control variables are significant and display plausible signs, which makes us confident that the information in SIGE do indeed measure the underlying economic variables of interest reasonably well. As one would expect, firms that expect to expand the workforce or that predict an increase in demand or that forecast to raise prices are more likely to report higher expected investment plans. Similarly, firms that perceive better current conditions of access to credit or better current conditions to invest are more likely to indicate higher investment expenditure plans. The marginal effects of the assessment of the current macroeconomic outlook in Italy or of the probability of an improvement are insignificant.

thus re-calculate the marginal effects changing the values of inflation expectations in the range from -0.2 to 2.8 percent (values that correspond to the 5th and 95th percentiles of the empirical distributions of inflation expectations of firms) while keeping the other controls at their baseline levels. The results shown in Appendix Table 4 indicate, perhaps not surprisingly, that the marginal effects of inflation expectations stay significant and virtually unaffected by the changing level of inflation expectations.

Another exercise we do is to assess the sensitivity of the marginal effects of inflation expectations when varying the evaluation levels of some other control variables, namely: the forecast of the expected labor demand in the next 3 months, the assessment of the current conditions of access to credit compared with previous 3 months, the assessment of the current economic outlook in Italy with respect to previous 3 months. We discussed earlier that these variables are trichotomous and typically the response categories depict worsening, stable or improving evaluations. Hence, we compute the marginal effects in three hypothetical cases obtained by evaluating the above set of variables *simultaneously* at the worse, neutral or better category and by also varying the evaluation level of inflation expectations in the coarser grid - 0.2, 0.0, 0.5, 0.9, 1.5 and 2.8 percent. The other explanatory variables are kept at the benchmark levels. Results are shown in Table 4.

Several remarks are in order. In all three cases, the marginal effects of inflation expectations on the predicted probability of expecting lower or higher investment expenditure plans are highly significant and display the same signs as in the baseline estimation. For any given level of expected inflation, the size of the marginal effects varies in the three scenarios. Let us consider first the scenario "Worsening assessment of idiosyncratic and macro situation" shown in Panel A. In this case, the marginal effects of inflation expectations are in absolute terms three to five times larger for the response category "lower" compared with the response category "higher". For instance, when expected inflation is evaluated at 0.9 percent the marginal effects of 12-month ahead inflation expectations on the predicted probabilities of reporting lower or higher investment expenditure plans are -0.045 and 0.014, respectively. They become -0.051 and 0.017 when using the 6-month ahead inflation expectations. Reversed results arise in the scenario "Improving assessment of idiosyncratic and macro situation" shown in Panel B. In this case, higher inflation expectations have larger effects on the response category "higher".

It is also interesting to consider the marginal effect of inflation expectations on the *net probability* of reporting higher investment expenditure plans, which we simply compute is as the difference between the marginal effects on the predicted probabilities of reporting higher and lower plans. Using the entries shown in Table 4, we find that the marginal effects of inflation expectations on the net probability attain the largest values in the stable scenario (about 0.07) and the smallest values in the improving scenario. Hence, these results point toward the existence of contingent effects of inflation expectations on the evaluation that firms have on their current and prospective economic situation. Furthermore, the fact that these effects are larger when firms perceive a worsening outlook than when they have a more buoyant evaluation means that the effectiveness of this expectations channel may be greater when a policy intervention is more needed.

6 Robustness checks

In this section we present the results of some robustness checks.

Original 5-category investment question

The first robustness check we conduct is to re-estimate the baseline specification (1) using the original formulation of the question on expected investment expenditure plans that admits five response categories. The estimation results shown in Panel A of Table 5 confirm all our previous findings. Regardless of the forecasting horizon, the marginal effects of inflation expectation on the predicted probability of reporting "much lower", "a little lower", "a little higher" or "much higher" expected investment plans are highly significant. In absolute terms, the effects are larger for the categories "a little lower" and "a little higher" compared with the respective boundary categories. The marginal effects on the category "about the same" are again weakly significant and negative. Quantitatively, though, these effects are very small.

Forecasting horizons of inflation expectations

Firms in SIGE report inflation expectations at horizons longer than one year ahead. Because investment decisions made today can have an impact on firm business over many years, it is worthy to assess to what extent longer-term inflation expectations affect the investment expenditure plans of firms. To this end, we re-estimate the baseline specification (1) using in turn the 2-year ahead inflation expectations and the 2-year ahead from 2 years inflation expectations and also using the 5-category dependent variable. The results are reported in Panel B of Table 5. Also in these cases, all our previous results are largely confirmed. The only notable difference regards the size of the marginal effects, which tend to become smaller with longer-term inflation expectations.

Time-varying estimation

As a further robustness check, we assess the time stability of the marginal effects of inflation expectations. To this end, we re-estimate the specification (1) augmented with interaction terms between the explanatory variables and the time fixed effects. Then, we evaluate the marginal effects at each point in time. The estimated marginal effects on the predicted probability of reporting lower or higher expected investment together with the 90 percent confidence interval (gray area) are presented in Figure 3. We find that the marginal effects of inflation expectations evaluated in the first three quarters of the sample period (until 2013Q2) are in general statistically insignificant and change signs compared with our previous results. In particular, in 2013Q1 and 2013Q2 firms with higher inflation expectations were more likely to lower their expected investment expenditure. This finding could then be rationalized with firms perceiving higher inflation expectations driven by a negative supply-shock. Since 2013Q4 the marginal effects of inflation expectations display the negative (or positive) sign for the predicted probability of reporting lower (or higher) investment expenditure and are in general statistically significant and rather stable over time. These results are consistent with firms considering higher inflation expectations to be driven by a positive demand shock or alternatively with the transmission of shocks when the economy is stuck at the effective lower bound. Theoretical work has shown that when at the effective lower bound (ELB) on policy rates the effect of demand shocks can be amplified, as shown for the case of fiscal shocks in Woodford (2011). Also a negative supply-side shock can have expansionary effects as the higher expected inflation induced by the shock lowers the ex-ante real rate thus stimulating interest-sensitive sectors of the economy and possibly offsetting the usual recessionary effects of the shock (Wieland, 2015).

Firm demographic characteristics

So far, all estimated marginal effects have been evaluated at the modal categories of the demographic characteristics of firms regarding the sector of economic activity, the geographical location and the degree of openness. Looking at the entries in Table 2 (rows (10)-(12)), this

means that we have computed the marginal effects of inflation expectations for a hypothetical firm that operates in the *industrial sector*, that is located in the North-West area of the country and that does not export. Yet, it could be that the effects of inflation expectations might differ along these observable characteristics of firms. Hence, we re-estimate specification (1) on sub-groups of firms. In order to maintain a sufficiently large sample size (say at least 2,000 observations) in the case of the geographical location and degree of openness we combine together the response categories "Centre" and "South & Islands" as well as the categories "34-66 percent" and "67-100 percent". The estimation results are shown in Table 6. In most cases, the marginal effects are significant and display the same signs as in the benchmark cases. By sectors of economic activity, we find that the marginal effects of inflation expectations are relatively larger for services firms than for industrial firms. Inflation expectations are positively related with the investment plans of construction firms but in this case the marginal effects are not significant. This later finding might depend on the small sample size used in the estimation. When looking at the geographical location of firms, we find that the largest marginal effects occur for firms that operate in the northern areas of Italy. Economic and social differences between the South and North of Italy have long been identified in the literature (Guiso et al., 2006). We also find smaller and less significant marginal effects of inflation expectations for firms that sell their products abroad, which likely reflects the fact that these exporters are less sensitive to business conditions in their home country since more of their revenues come from foreign sources.

7 Inspecting the channels

In previous sections we found robust evidence of a positive relationship between the inflation expectations of firms and their expected investment expenditure. But why do firms with higher inflation expectations raise more their investment expenditure plans? And are there firm characteristics that can affect such a relationship? To answer these questions we inspect three potential channels.

A fundamental tenet in modern investment theory posits that firms' investment expenditure is negatively related to the ex-ante real interest rate. *Ceteris paribus*, a firm that expects a higher inflation rate perceives a lower *ex-ante* real interest rate and thus has an incentive to increase investment expenditure. To investigate the relevance of this *interest rate*

channel we modify the baseline latent specification (1) in three ways. In one case, we replace the inflation expectation of firms at horizon h with the firm-level *ex-ante* real interest rate at the same horizon (RINT_t^{*i,h*}) that is:

$$INVEST_t^{i,*} = \beta RINT_t^{i,h} + \gamma \mathbf{X}_t^i + error_t^i.$$
⁽²⁾

Note that in specification (2) we are implicitly restricting the coefficient on the nominal interest rate and the coefficient on inflation expectation to sum to zero. There are reasons to believe that this restriction might be violated if, for example, firms are allowed to fiscally deduct the nominal interest payments or if there are mechanisms beyond the interest rate channel through which inflation expectations affect the willingness to invest of firms. Furthermore, estimation of a latent specification like (2) does not allow us to single out the effects of inflation expectations.

These considerations lead us to consider an alternative specification in which the inflation expectations of firms and the nominal interest rate (NINT^{*i*}_{*t*}) enter the regression as separate covariates, that is

$$INVEST_t^{i,*} = \beta_1 INFL_t^{i,h} + \beta_2 NINT_t^i + \gamma \mathbf{X}_t^i + error_t^i.$$
(3)

Unfortunately, both regression specifications (2) and (3) share a common drawback that has to do with the fact that when using the nominal interest rate on new term loans we obtain a large number of missing observations. As shown in Table 2, the number of observations of the nominal rate is about 3,400 and remains virtually unaffected when constructing the ex-ante real interest rates. Hence, to maximize the sample size we further modify the latent specification by replacing the firm-level *quarterly* nominal interest rate with the firm-level *annual* score lagged by one year (SCORE $_{t-4}^{i}$), that is

$$INVEST_t^{i,*} = \beta_1 INFL_t^{i,h} + \beta_2 SCORE_{t-4}^i + \gamma \mathbf{X}_t^i + error_t^i.$$
(4)

Two remarks regarding the above specification are in order. First, we use the score lagged by one year to capture the fact that in reality this indicator is made available with a delay of about 15

months. Thus, in the four quarters of each year we repeat the value of the score referred to the previous year. Second, the inclusion of the score in the specification is specifically meant to track the component of the nominal borrowing cost related to the default risk of firm.¹⁵ The other main component of the nominal borrowing cost, being the market risk-free rate, is indirectly accounted for by the time fixed effects.

Beyond the traditional interest rate channel, expected inflation can affect the willingness of firms to invest by interacting with their balance sheet characteristics. As discussed in the Introduction, on the one hand higher expected inflation can erode the expected real value of liquid assets generating a *negative* wealth effect that in turn may discourage investment expenditure. On the other hand, higher inflation expectations may induce firms to adjust the composition of their assets substituting away from liquid assets into capital goods, generating in this way a Tobin effect.

Anticipated inflation can also interact with the liabilities of firms giving rise to opposing effects. On the one hand, higher expected inflation generates a *positive* wealth effect on firms as it erodes the expected nominal value of outstanding debt. This, in turn, may create additional borrowing capacity and lead firms to raise their investment expenditure. On the other hand, if higher future inflation is accompanied by future higher nominal interest rates than highly indebted firms might have to pay higher interest rate expenses and thus be discouraged from undertaking investment expenditure plans.

To assess whether the liquidity and debt positions of firms alter the effects of inflation expectations we modify the latent specification (4) as follows:

$$INVEST_{t}^{i,*} = INFL_{t}^{i,h} (\beta_{1p} + \beta_{2p} Z_{t-4}^{i}) + SCORE_{t-4}^{i} (\beta_{1s} + \beta_{2s} Z_{t-4}^{i}) + + \mathbf{X}_{t}^{i} (\beta_{1x} + \beta_{2x} Z_{t-4}^{i}) + \beta_{3} Z_{t-4}^{i} + error_{t}^{i},$$
(5)

where Z_{t-4}^{i} represents either the annual cash ratio (CASH_{t-4}^{i}) or debt ratio (DEBT_{t-4}^{i}) of firm lagged by one year. As with score, in the four quarters of each year we use the values of the cash and debt ratios recorded in the previous year. Furthermore, we use the one-year lagged values of

¹⁵ Note that types of risk premia possibly related to the economic sectors in which firms operate or the geographical areas in which they are located are accounted for by the inclusion of firm demographic characteristics.

the ratios to overcome the endogeneity problem between the inflation expectations of firms and their decisions to adjust the balance-sheet composition.

Note that we let the balance-sheet ratios enter the specification as individual regressors and also interacted with *all* control variables (except for the time fixed effects). In particular, it is thanks to the term of interaction between expected inflation and the balance-sheet ratio that we can trace the effect of inflation expectation for different values of firm liquidity or debt. The partial effect of higher inflation expectation on the investment expenditure plans of firms is given by $\beta_{1p} + \beta_{2p} Z_{t-4}^i$.

7.1 Results

In this section we present the estimation results obtained from the specifications (2)-(5). We report the estimated coefficients as well as the marginal effects on the predicted probability to expect lower or higher investment expenditure plans only for a subset of regressors, namely the inflation expectation, the nominal interest rate, the score, the cash ratio, the debt ratio and the interaction terms if included in the specification.

The interest rate channel

The estimation results for the investigation of the interest rate channel are reported in Table 7.¹⁶ To begin with, in columns (1), (5) and (6) we show the estimation results of specification (2) and find evidence of a highly significant and contractionary effect of the *ex-ante* real interest rate. Thus, consistent with the theory, a higher real interest rate increases (decreases) the probability that firms indicate lower (higher) investment expenditure plans. Quantitatively, the marginal effects are rather small (about 0.01 and -0.01, respectively). In columns (2), (7) and (8) we report the estimation results obtained using specification (3), in which inflation expectation and the nominal borrowing cost are treated as separate regressors. In this case, once controlling for the nominal borrowing cost, the effects of inflation expectations are (weakly) significant and display the expected signs, confirming that inflation expectations of firms are positively related with their willingness toward investment expenditure. Using the 12-month ahead inflation expectations the marginal effects on the predicted probability of reporting lower

¹⁶ In Appendix Tables 9 and 10 we report the estimated marginal effects of all the control variables, except for the sector and time fixed effects (that are available upon request).

or higher investment expenditure plans are -0.024 and 0.030, respectively. The effects are somewhat larger when we use the 6-month ahead expectations.

Next, in columns (3), (9) and (10) we show the estimation results obtained using specification (4), which features the score in place of the nominal borrowing cost, and the same set of observations as in the previous two cases.¹⁷ Interestingly, the estimated coefficient of the score is highly significant and displays the expected signs. Furthermore, the marginal effects of inflation expectations are virtually unchanged compared with the results shown in columns (3) and (4). Finally, in columns (4), (11) and (12) we report the results obtained re-estimating specification (4) using all the available observations (the sample rises from about 4,300 to 9,300 observations). It is worth noting that also in this case all previous findings are confirmed and moreover the marginal effects of inflation expectations return to be highly significant.

The cash channel

We now turn to the cash channel through which inflation expectations may affect firm economic behavior. The estimated coefficients results obtained using specification (5) are reported in Panel A of Table 8 while the marginal effects are shown in Table 9. The latter are calculated for different levels of the cash ratio ranging zero to 60 percent (values that correspond to the 10th and 90th percentile of the full-sample empirical distribution of cash ratio).

Several results are worth noting. We find that the estimated coefficients of inflation expectation remain highly significant and positive, with magnitudes that are virtually identical to those reported in Table 7. More interestingly, though, the estimated coefficient of the interaction term between inflation expectation and the cash ratio is positive (at most weakly significant with the 6-month ahead inflation expectation) indicating larger expansionary effects of inflation expectation for more liquid firms. As shown in Table 9, these results are confirmed when we calculate the marginal effects, which remain highly significant and more importantly exhibit (in absolute terms) an increasing relationship with the evaluation levels of the cash ratio. For example, the marginal effect of the 12-month ahead inflation expectation on the predicted probability of reporting higher investment is 0.035 when the cash ratio is set at 5 percent and increases to 0.056 when the cash ratio is at 60 percent. Somewhat weaker marginal effects are

¹⁷ For simplicity, we treat the score as if it were a continuous variable. Estimation results do not change if we treat it as a 9-category ordered qualitative variable (results are available upon request).

obtained on the predicted probability of reporting lower investment expenditure plans. The estimated marginal effects are relatively stronger when using the 6-month ahead inflation expectations. In sum, these findings reveal that the expansionary effects of higher expected inflation are larger for firms that hold a great amount of liquid assets.

Further interesting results emerge with reference to the other selected variables. As shown in Table 8, we find that the highly statistically significant negative effect of the score (roughly equal to -0.05) is attenuated by the cash ratio as indicated by the estimated coefficient of the interaction term between the score and the cash ratio that is positive and significant at 5 percent. The marginal effects of the score confirm this finding thus suggesting that the contractionary effects of the score on firm investment expenditure plans lessens for higher levels of the cash ratio. When the value of the cash ratio is equal to 60 percent the marginal effect of the score becomes statistically insignificant. These results can be explained by the fact that firms with a large amount of liquid assets can more easily finance investment expenditure with internal resources, thereby being less sensitive to the interest rate (Sharpe and Suarez, 2013). Lastly, the marginal effects of the cash ratio display the correct signs, i.e. higher level of firm liquidity is associated with a decrease (increase) in the predicted probability of reporting lower (higher) expected investment, and are in general statistically significant.

The debt channel

To assess the debt channel we report the estimation results obtained using specification (5) in panel B of Table 8 and in Table 10. We calculate the marginal effects letting the levels of the debt ratio range between 20 to 80 percent (values that correspond to the 10^{th} and 90^{th} percentile of the full-sample empirical distribution of the debt ratio).

Also in this case several interesting results emerge. Looking at the entries shown in panel B of Table 8, we find that the estimated coefficients of inflation expectation are not statistically insignificant whereas the estimated coefficients of the interaction terms between the inflation expectation and the debt ratio are significant and positive. This result suggests two things. The first one is that the magnitude of estimated coefficient of inflation expectation on firm investment plans magnitude is increasing in the level of the debt ratio thus providing support to the fact that higher expected inflation may create a positive wealth effect and generate additional borrowing capacity. The second thing, though, is that for low levels of the debt ratio the overall

estimated coefficient of inflation expectation might become statistically insignificant. In this case, the positive wealth effects induced by expected inflation are small and likely offset by the recessionary effects that higher inflation expectations may bring about. Interestingly, the marginal effects of inflation expectations reported in Table 10 confirm all these results. For instance, the marginal effect of the 12-month ahead inflation expectations on the predicted probability of reporting higher investment expenditure plans nearly triplicates (from 0.028 to 0.074) when the debt ratio rises from 30 to 80 percent. Yet, the marginal effect becomes statistically insignificant (though still displaying a positive sign) when the debt ratio is set to 20 percent. When using the 6-month ahead inflation expectations, we find similar results with the only difference that the marginal effect of inflation expectation remains significant also for the lowest level of the debt ratio.

Finally, turning to the other regressors we find that the estimated coefficient of the score is statistically significant and negative (about -0.045) while that of the interaction term between the score and the debt ratio is statistically insignificant suggesting the lack of any relationship with the debt ratio. This evidence is also confirmed by the marginal effects of the score that in all cases appear virtually unaffected by the values of firm indebtedness. Finally, the estimated coefficients and the marginal effects of the debt ratio are statistically insignificant.

Recap of results

The results presented in this section indicate that firms' inflation expectations are positively related with their investment expenditure plans through multiple channels. First of all, we document the working of the conventional *ex-ante* real interest rate channel. Once we control for the firm-level nominal borrowing cost (directly measured by the nominal interest rate charged on new term loans or proxied by the default risk score) we find that firms that expect higher inflation, and thus a lower ex-ante real interest rate, are more likely to raise or reduce less their investment expenditure plans. Beyond this, the positive relationship between inflation expectations and firm investment expenditure plans also depends on the levels of liquidity and debt of firms. In particular, the estimated marginal effects of inflation expectations are stronger for more liquid or more indebted firms. This is consistent with the view that higher expected inflation generates two effects: 1) it make firms more willing to spend (invest) their cash holding to escape the negative wealth effect on assets and 2) it erodes the nominal value of outstanding

debt thus creating additional borrowing capacity. That said, we also find that the expansionary effects of inflation expectations may vanish (become statistically insignificant) for low enough levels of the debt ratio.

8 Conclusions

In this paper we use Italian survey data to provide new evidence on the nexus between the inflation expectations of firms and their willingness to invest. Estimating a series of order probit regressions over the period 2012Q4-2016Q4, we find that such a relationship exists and is positive, suggesting that higher expected inflation is associated with a larger willingness towards investment expenditure. To investigate the channels through which inflation expectations may affect firm economic decisions we merge the survey data with two other data sources to obtain information on the nominal borrowing cost of firms and some balance sheet characteristics. Using this information we document several relevant results. First, we provide empirical support to the working of the standard interest rate channel according to which higher expected inflation lowers the *ex-ante* real interest rate and thus stimulates investment expenditure. Second, we find significant interactions between the inflation expectations of firms and their balance sheet characteristics, namely the liquidity and debt position. Our results indicate that the magnitude of the expansionary effects of higher expected inflation becomes larger when firms hold a large amount of liquid assets and/or are more indebted.

In general, these results lend support to the policy measures adopted in the aftermath of the Great Recession by central banks that operating through the expectations channels were aimed at engineering higher inflation expectations to stimulate the economy. Furthermore, the fact that the effects of higher inflation expectations appear to vary with the liquidity and debt position of firms suggests the existence of synergic forces that policy-makers might exploit when engineering policies aimed at increasing inflation expectations (like for instance the forward guidance announcements) and policies aimed at channeling funds to firms (like for instance the targeted longer-term refinancing operations adopted by the European Central Bank). The potential benefits from the expectations channel can be large, it remains an open question how policy-makers could effectively drive agents' inflation expectations.

References

- Altman, E. (1968). Financial Ratios, Discriminant Analysis and the Prediction of Corporate Bankruptcy. *Journal of Finance*, 21, 589–609.
- Altman, E., G. Marco and F. Varetto (1994). Corporate Distress Diagnosis: Comparisons Using Linear Discriminant Analysis and Neural Networks (the Italian Experience). *Journal of Banking and Finance*, 18, 505–29.
- Bachmann, R., T. O. Berg and E. R. Sims (2015). "Inflation Expectations and Readiness to Spend: Cross-Sectional Evidence." *American Economic Journal: Economic Policy* 7 (1), 1–35.
- Bachmann, R. and S. Elstner (2015). "Firm optimism and pessimism." *European Economic Review* 79, 297–325.
- Bartiloro, L., M. Bottone and A. Rosolia (2017). What does the heterogeneity of the inflation expectations of Italian firms tell us? Banca d'Italia Working Paper 414.
- Bryan, M. F., B. H. Meyer and N. B. Parker (2015). The inflation expectations of firms: What do they look like, are they accurate, and do they matter? Federal Reserve Bank of Atlanta, Working paper 2014-27a.
- Burke, M. A. and A. K. Ozdagli (2013). Household inflation expectations and consumer spending: evidence from panel data. Federal Reserve Bank of Boston Research Department Working Papers No. 13-25.
- Cloyne, J., L. Boneva, M. Weale and T. Wieladek (2016). "Firms' expectations and pricesetting: Evidence from micro data." External MPC Unit Discussion Paper, No. 48, Bank of England.
- Coibion, O., Y. Gorodnichenko, S. Kumar and M. Pedemonte (2018). Inflation Expectations A Policy Tool? Proceedings of the ECB Forum on Central Banking.
- Coibion, O., Y. Gorodnichenko and T. Ropele (2018). Inflation Expectations and Firm Decisions: New Causal Evidence. Mimeo, Banca d'Italia.
- Conflitti, C. and R. Zizza (2018). What's behind firms' inflation forecasts? Mimeo, Banca d'Italia.
- Crump, R. K., S. Eusepi, A. Tambalotti and G. Topa (2015). "Subjective Intertemporal Substitution," Federal Reserve Bank of New York, Staff Paper Number 734.

- D'Acunto, F., D. Hoang and M. Weber (2018). "Unconventional Fiscal Policy." NBER Working Paper 24244.
- Dräger, L. and G. Nghiem (2016). "Are Consumers' Spending Decisions in Line With an Euler Equation?" Gutenberg School of Management and Economics, Working Papers 1802.
- Duca, I. A., G. Kenny and A. Reuter (2017). Inflation Expectations, Consumption and the Lower Bound: Empirical Evidence from a Large Micro Panel. Manuscript.
- Eggertsson, G. B. (2008). Great Expectations and the End of the Depression. *American Economic Review* 98(4), 1476-1516.
- Eggertsson, G. B. and M. Woodford (2003). The Zero Bound on Interest Rates and Optimal Monetary Policy. *Brookings Papers on Economic Activity*, 34(1), 139-235.
- Guiso, L., P. Sapienza and L. Zingales (2006). "Does Culture Affect Economic Outcomes?" Journal of Economic Perspectives, 20(2), 23-48.
- Ichiue, H. and S. Nishiguchi (2015). Inflation Expectations and Consumer Spending at the Zero Bound: Micro Evidence. *Economic Inquiry*, 53, 1086–1107.
- Kumar, S., H. Afrouzi, O. Coibion and Y. Gorodnichenko (2015). Inflation Targeting Does Not Anchor Inflation Expectations: Evidence from Firms in New Zealand. *Brookings Papers* on Economic Activity, 46(2), 151-225.
- Panetta, F., F. Schivardi and M. Shum (2009). Do Mergers Improve Information? Evidence from the Loan Market. *Journal of Money, Credit and Banking*, 41(4), 673–709.
- Richards, S. and M. Verstraete (2016). Understanding firms' inflation expectations using the Bank of Canada's Business Outlook Survey. Bank of Canada Staff Working Paper 7.
- Ropele, T. (2018). Inflation expectations and price setting behavior: evidence from business survey data. Mimeo, Banca d'Italia.
- Sharpe, S. A. and G. A. Suarez (2013). "The insensitivity of investment to interest rates: Evidence from a survey of CFOs." Finance and Economics Discussion Series 2014-2, Board of Governors of the Federal Reserve System (U.S.).
- Wieland, J. F. (2015). "Are Negative Supply Shocks Expansionary at the Zero Lower Bound?" Tech. rep.
- Woodford, M. (2011). "The Simple Analytics of the Fiscal Multiplier," American Economic Journal: Macroeconomics, 3(1), 1-35.

Figures and Tables



Figure 1. Expected Investment: Frequencies of Response Over Time.

Notes. In panels A to E we show the time development of the relative frequency of responses (5 categories) to the SIGE question that asks firms about their expected investment expenditure. In Panel F we show the net percentage of higher expected investment expenditure (calculated as the difference between the sum of relative frequencies of responses "much higher" and "a little higher" and the sum of relative frequency of responses "much lower" and "a little lower") together with the actual (annual) growth rates of non-residential investment expenditure in Italy in nominal terms (red line) and real terms (green line). In correspondence to 2013Q3 the histogram is left blank as in that survey wave the question on expected investment expenditure was not presented to firms. All statistics are computed with sampling weights. Values are expressed in percentage.



Figure 2. Inflation Expectations: Developments Over Time.

Notes. In panels A and B we plot the time development of the mean value and standard deviation of firms' inflation expectations (12-month and 6-month ahead). In panels C and D we plot the mean values of firms' inflation expectations (12-month and 6-month ahead) distinguishing between firms that reported to expect lower or higher investment expenditure. In correspondence to the third quarter of 2013 the histogram is left blank as in that survey wave the question on expected investment expenditure was not presented to firms. All statistics are calculated using sampling weights. Values are expressed in percentages.



Figure 3. Time-varying Estimation of Marginal effects of Inflation Expectations

Notes. In panels A and B we plot the time-varying marginal effects of firms' inflation expectations (12-month and 6-month ahead) on the predicted probability to report lower or higher expected investment expenditure. The grey areas represent the 90 percent confidence band. In correspondence to the third quarter of 2013 the marginal effect is left blank as in that survey wave the question on expected investment was not presented to firms.

Question	Response								
Your firm's total number of employees in the next 3 months will be	Lower, Unchanged, Higher								
How will the total demand for your products vary in the next 3 months?	Decrease, No change, Increase								
For the next 12 months, what do you expect will be the average change in your firm's prices?	Percentage points with one decimal digit								
Please indicate direction and intensity of the following factors as they will affect your firm's selling prices in the next 12 months: - raw material prices - labor cost	Downward high, Downward average, Downward low, Neutral, Upward low, Upward average, Upward high								
Compared with 3 month ago, do you think conditions for investment are?	Worse, The Same, Better								
Compared with three months ago, are credit conditions for your company?	Worse, Unchanged, Better								
Compared with 3 months ago, do you consider Italy's general economic situation is?	Worse, The same, Better								
What do you think is the probability of an improvement in Italy's general economic situation in the next 3 months?	Zero, 1-25%, 26-50%, 51-75%, 76-99%								
		Obs.	ŀ	Relative frequ	iency of	each res	ponse c	ategory	7
------	--	--------	------	----------------	----------	----------	---------	---------	------
Row	Qualitative variables			(percenta	ge value	es)		
			(a)	(b)	(c)	(d)	(e)	(f)	(g)
((1)	Expected investment expenditure	10,563	8.3	15.5	48.0	24.5	3.7		
(2)	Own-product expected demand	10,436	13.7	63.9	22.4				
(3)	Expected employment	10,486	19.5	68.5	12.0				
(4)	Labor cost for expected price change	10,284	0.8	1.8	2.9	58.2	16.5	14.2	5.5
(5)	Price of raw materials for exp. price change	10,286	1.0	3.1	5.4	50.1	19.7	16.0	4,7
(6)	Current conditions to invest	10,457	19.0	70.2	10.8				
(7)	Access conditions to credit	10,370	15.7	76.6	7.7				
(8)	Italy's macroeconomic outlook	10,432	24.9	63.9	11.2				
(9)	Prob. of improvement of Italy's outlook	10,513	36.2	47.5	10.7	5.0	0.6		
(10)	Sector of economic activity	10,563	41.6	38.7	19.8				
(11)	Geographical area	10,563	37.5	26.4	18.5	17.7			
(12)	Share of revenues from exports	10,563	46.6	22.6	17.7	13.1			
Darr	Overtiteting engickles	Obs.	Mean	Std. Dev.	p10	p25	p50	p75	p90
KOW	Quantitative variables			(percenta	ge value	es)	•	•
(13)	Number of employees (in log)	10,563	4.71	0.88	3.91	3.93	4.51	5.12	5.87
(14)	Inflation expectation (12m)	10,475	0.84	0.84	0.00	0.20	0.60	1.10	2.20
(15)	Inflation expectation (6m)	10,475	0.71	0.85	0.00	0.10	0.50	1.00	2.20
(16)	Own-price expected growth rate	10,563	0.42	5.20	-2.00	0.00	0.00	1.50	3.00
	Interest rate on new term loans:								
(17)	Nominal	3,407	3.62	2.20	1.05	1.94	3.25	4.98	6.71
(18)	Ex-ante real (12m)	3,384	2.79	2.22	0.27	1.16	2.39	4.11	5.84
(19)	Ex-ante real (6m)	3,384	2.91	2.21	0.42	1.28	2.53	4.23	5.97
(20)	Cash ratio	7,733	19.7	32.9	0.20	1.10	5.40	23.5	57.8
(21)	Debt ratio	7,688	50.1	22.3	20.2	32.3	48.8	67.6	79.0

Table 2. Basic Descriptive Statistics Of Selected SIGE Variables.

Notes. In this table we report basic descriptive statistics of selected SIGE variables. Statistics are computed on pooled data over the period from 2012Q4 to 2016Q4 using sampling weights. For the qualitative variables we report the number of observations and the relative frequency of response categories, which are: (1)-(a) "much lower", (1)-(b) "a little lower", (1)-(c) "about the same", (1)-(d) "a little higher", (1)-(e) "much higher"; (2)-(a) "decrease", (2)-(b) "no change", (2)-(c) "increase"; (3)-(a) "lower", (3)-(b) "unchanged", (3)-(c) "higher"; (4)-(a) "downward high", (4)-(b) "downward average", (4)-(c) "downward low", (4)-(d) "neutral", (4)-(e) "upward low", (4)-(f) "upward average", (4)-(g) "upward high"; (5)-(a) "downward high", (5)-(b) "downward average", (5)-(c) "downward low", (5)-(d) "neutral", (5)-(e) "upward low", (5)-(f) "upward average", (5)-(g) "upward high"; (6)-(a) "worse", (6)-(b) "the same", (6)-(c) "better"; (7)-(a) "worse", (7)-(b) "unchanged", (7)-(c) "better"; (8)-(a) "worse", (8)-(b) "the same", (8)-(c) "better"; (9)-(a) "zero", (9)-(b) "1-25%", (9)-(c) "26-50%", (9)-(d) "51-75%", (9)-(e) "76-99%"; (10)-(a) "industry", (10)-(b) "services", (10)-(c) "construction"; (11)-(a) "North-West", (11)-(b) "North-East", (11)-(c) "Centre", (11)-(d) "South & Islands"; (12)-(a) "zero", (12)-(b) "1-33 percent", (12)-(c) "34-66 percent", (12)-(d) "67-100 percent". For the quantitative variables we report the number of observations, the mean, the standard deviation and the 10th, 25th, 50th and 75th and 90th percentiles. Firm-level ex-ante real interest rates are calculated as the difference between the nominal interest rate and expected inflation. The cash ratio is calculated as the sum of cash and marketable securities over current liabilities (i.e. debts due within one year) while the debt ratio is the amount of current liabilities over total assets. All statistics are calculated using sampling weights.

	Ordered pro	bit estimation	
Estimated coefficient	Marginal effec	ts on predicted proba	ability to invest
	lower	about the same	higher
(1)	(2)	(3)	(4)
0.116*** (0.028)	-0.033*** (0.008)	-0.006** (0.003)	0.038*** (0.009)
9,615			
0.083			
0.131*** (0.032)	-0.037*** (0.009)	-0.006** (0.003)	0.043*** (0.011)
9,615 0.083			
	Estimated coefficient (1) 0.116*** (0.028) 9,615 0.083 0.131*** (0.032) 9,615 0.083	Ordered pro Estimated coefficient Marginal effect 10wer 10wer (1) (2) 0.116*** -0.033*** (0.028) (0.008) 9,615 0.032) 0.131*** -0.037*** (0.032) -0.037*** 9,615 0.009)	Ordered probit estimation Estimated coefficient Marginal effects on predicted probation lower about the same (1) (2) (3) 0.116*** -0.033*** -0.006** (0.028) -0.037*** -0.006** 0.131*** -0.037*** -0.006** (0.032) -0.037*** -0.006** 9,615 0.003) (0.003)

Table 3. Effects of Inflation Expectations on Investment Expenditure Plans: Baseline Results

Notes. In this table we report the estimation results of specification (1) to study the effects of inflation expectations of firms on their investment expenditure plans. We only report the estimated coefficient (column (1)) as well as the estimated marginal effects (columns (2)-(3)) of inflation expectations. The marginal effects are computed using the sample mean values for the continuous explanatory variables and the sample modal categories for the qualitative explanatory variables. Estimates of the other controls (except for the time fixed effects) are reported in Appendix Tables 2 and 3. ***, **, * denote statistical significance at 1, 5 and 10 percent level. Robust standard errors are in parentheses.

		Eval	uation levels	of inflation ϵ	expectations	and response	e categories to	o question on	i investment e	xpenditure p	lans	
	-0.2 p	ercent	0.0 pe	ercent	0.5 pc	ercent	0.9 p	ercent	1.5 pe	rcent	2.8 pe	rcent
Regressor	lower	higher	lower	higher	lower	higher	lower	higher	lower	higher	lower	higher
	(1)	(2)	(3)	(4)	(2)	(9)	(1)	(8)	(6)	(10)	(11)	(12)
				Panel A.	Worsening a	assessment o	f idiosyncrati	c and macro	situation			
INFL ^{e,12m}	-0.044^{***}	0.012***	-0.044^{***}	0.012***	-0.045***	0.013***	-0.045***	0.014^{***}	-0.045***	0.016^{***}	-0.045^{***}	0.019***
INFL ^{e,6m}	(0.010)-0.050***	(0.003) 0.013^{***}	(0.010)-0.050***	(0.003) 0.014^{***}	(0.011) -0.051***	(0.003) 0.015^{***}	(0.011) -0.051***	(0.004) 0.017^{***}	(0.011)-0.051***	(0.005) 0.019^{***}	(0.010) -0.051***	(0.006) 0.023***
	(0.012)	(0.003)	(0.012)	(0.003)	(0.012)	(0.004)	(0.012)	(0.005)	(0.012)	(0.006)	(0.012)	(0.008)
				Panel B.	Improving a	ssessment of	f idiosyncrati	c and macro	situation			
INFL ^{e,12m}	-0.010^{***}	0.043^{***}	-0.010***	0.042^{***}	-0.009***	0.042^{***}	-0.008***	0.041^{***}	-0.007***	0.039^{***}	-0.005***	0.036***
INFI e.6m	(0.003)	(0.011) 0.048***	(0.003)	(0.010) $0.048***$	(0.003) -0.010***	(0.010) 0.047***	(0.002)-0.009***	(0.010) 0.046^{***}	(0.002)-0.008***	(0.009) 0.044***	(0.001)	(0.007) 0.040***
	(0.004)	(0.012)	(0.004)	(0.012)	(0.003)	(0.011)	(0.003)	(0.011)	(0.002)	(0.010)	(0.001)	(0.008)
				Panel	C. Stable ass	essment of i	diosyncratic	and macro sit	tuation			
INFL ^{e,12m}	-0.035***	0.035***	-0.035***	0.036^{***}	-0.033***	0.037^{***}	-0.032***	0.038^{***}	-0.030***	0.040^{***}	-0.026***	0.042^{***}
INTET e,6m	(0.009) 0.000***	(0.008) 0.040***	(0.009) 0.020***	(0.008)	(0.008)	(0.00) 0.002***	(0.008) 0.026***	(0.009) 0.044***	(0.007) 0.024***	(0.010)	(0.005) 0.020***	(0.011)
INFL	-0.040	(0.09)	(0.010)	(0.09)	(600.0)	(0.010)	(600.0)	(0.011)	(0.007)	(0.012)	-0.028	(0.013)
Notes. In expenditury	this table we plans. The plans of the plans of the plane	report the marginal eff	marginal effe	ects of infla ned using sp	tion expecta ecification (tions of firn 1) and evalu	ns on the pr ated for diffe	edicted prob erent levels c	ability of rej of inflation ex	porting lowe	r or higher a the grid -0.	investment 2, 0.0, 0.5,
0.9, 1,5 and	1 2.8 percent.	. Furthermor	e, in Panel A	"Worsening	assessment	of idiosyncr	atic and mac	ro situation"	the marginal	effects are c	alculated ass	uming that

firms expect in the next 3 months a lower labor demand and that with respect to the previous 3 months indicate worse current conditions of access to credit, worse conditions to invest and a worse evaluation of Italy's economic outlook. In Panel B "Improving assessment of idiosyncratic and macro situation" the marginal effects are calculated assuming that firms expect in the next 3 months a higher labor demand and that with respect to the previous 3 months indicate better current conditions of access to credit, better conditions to invest and a better evaluation of Italy's economic outlook. Finally, in Panel C "Stable assessment of idiosyncratic and macro situation" the marginal effects are calculated assuming that firms expect in the next 3 months a stable labor demand and that with respect to the previous 3 months indicate stable current conditions of access to credit, stable conditions to invest and a stable evaluation of Italy's economic outlook. All the other regressors (expect for inflation expectations) are evaluated at benchmark levels as in Tables 2 and 3. *, **, and *** denote statistical significance at the 10%, 5% and 1% critical level, respectively. Robust standard errors are in parenthesis.

	Margina	al effects of or	n predicted p	robability to	invest	Obs.	Pseudo R-square
	much lower	a little lower	about the same	a little higher	much higher		
Regressors	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Panel A.							
Inflation expectations 6-month ahead	-0.013***	-0.019***	-0.007**	0.030***	0.008***	9,615	0.070
Inflation expectations 12-month ahead	(0.003) -0.010*** (0.003)	(0.005) -0.016*** (0.004)	(0.003) -0.006** (0.002)	(0.008) 0.025*** (0.007)	(0.002) 0.007*** (0.002)	9,615	0.070
Panel B.							
Inflation expectations 2-year ahead	-0.008*** (0.002)	-0.012*** (0.003)	-0.004** (0.002)	0.019*** (0.005)	0.005*** (0.001)	9,615	0.070
Inflation expectations 2-year ahead from 2 years	-0.006*** (0.002)	-0.009*** (0.003)	-0.005** (0.002)	0.016*** (0.005)	0.004*** (0.001)	7,332	0.064

Table 5. Robustness Check: Five-Category Dependent Variable and Forecasting Horizons

Notes. In this table we report the marginal effects of inflation expectations on the predicted probability of reporting one of the five categories for expected investment expenditure using specification (1). In Panel A we use in turn the 12-month and the 6-month ahead inflation expectations while in Panel B we use longer-term inflation expectations. The marginal effects are computed using the full-sample mean values for the continuous explanatory variables and the full-sample mode categories for the qualitative explanatory variables. Estimates for the other controls are reported in Appendix Tables 5-8. ***, **, * denote statistical significance at 1, 5 and 10 percent level. Robust standard errors are in parentheses.

	Marginal effect	cts of on predicted proba	bility to invest	Obs.	Pseudo R-square
—	lower	about the same	higher		
-	(1)	(2)	(3)	(6)	(7)
Panel A. Sector of economic activity					
"Industry"					
Inflation expectations 6-month ahead	-0.030**	-0.010*	0.040**	3,924	0.108
-	(0.013)	(0.005)	(0.017)		
Inflation expectations 12-month ahead	-0.024**	-0.008*	0.033**	3,924	0.108
	(0.011)	(0.005)	(0.015)		
"Services"					
Inflation expectations 6-month ahead	-0.058***	-0.003	0.061***	3,878	0.054
	(0.016)	(0.004)	(0.018)		
Inflation expectations 12-month ahead	-0.052***	-0.002	0.054***	3,878	0.054
" <u>c</u> . "	(0.014)	(0.004)	(0.015)		
Construction	0.010	0.007	0.010	1.012	0.101
Inflation expectations 6-month ahead	-0.012	-0.006	0.018	1,813	0.121
Inflation consists (is no 10 month all and	(0.014)	(0.007)	(0.021)	1 0 1 2	0.121
Inflation expectations 12-month anead	-0.011	-0.006	0.017	1,813	0.121
Panel B. Geographical location	(0.012)	(0.006)	(0.018)		
"North-West"					
Inflation expectations 6-month ahead	-0 037**	-0.005	0.042*	2 744	0.315
initiation expectations o-month allead	(0.019)	(0.003)	(0.042)	2,744	0.515
Inflation expectations 12-month ahead	-0.032**	-0.004	0.037**	2 744	0.315
minuton expectations 12 month aread	(0.052)	(0.003)	(0.05)	2,744	0.515
"North-East"	(0.010)	(0.000)	(0.010)		
Inflation expectations 6-month ahead	-0.050***	-0.006	0.057***	2.661	0.056
r	(0.018)	(0.004)	(0.020)	_,	
Inflation expectations 12-month ahead	-0.037**	-0.005	0.041**	2,661	0.055
1 I	(0.015)	(0.003)	(0.018)	,	
"Centre and South & Islands"			· · · ·		
Inflation expectations 6-month ahead	-0.023**	-0.002*	0.025**	4,210	-0.123
-	(0.012)	(0.001)	(0.013)		
Inflation expectations 12-month ahead	-0.025**	-0.002*	0.027**	4,210	-0.123
-	(0.010)	(0.001)	(0.011)		
Panel C. Share of revenues from exports					
"Zero"					
Inflation expectations 6-month ahead	-0.042***	-0.006**	0.048***	4,498	0.074
× ×	(0.013)	(0.003)	(0.015)		
Inflation expectations 12-month ahead	-0.036***	-0.005**	0.041***	4,498	0.074
*	(0.011)	(0.002)	(0.013)		
"1-33 percent"	•				
Inflation expectations 6-month ahead	-0.019	-0.006	0.025	2,083	0.118
	(0.018)	(0.006)	(0.023)		
Inflation expectations 12-month ahead	-0.018	-0.006	0.023	2,083	0.118
	(0.016)	(0.005)	(0.021)		

		Continu	es from pre	vious page
-0.036**	-0.008**	0.044**	3,034	0.087
(0.015)	(0.004)	(0.018)		
-0.030**	-0.007**	0.036**	3,034	0.087
(0.013)	(0.003)	(0.016)	,	
	-0.036** (0.015) -0.030** (0.013)	-0.036**-0.008**(0.015)(0.004)-0.030**-0.007**(0.013)(0.003)	-0.036** -0.008** 0.044** (0.015) (0.004) (0.018) -0.030** -0.007** 0.036** (0.013) (0.003) (0.016)	-0.036** -0.008** 0.044** 3,034 (0.015) (0.004) (0.018) -0.030** -0.007** 0.036** 3,034 (0.013) (0.003) (0.016) 0.016)

Notes. In ***, **, * denote statistical significance at 1, 5 and 10 percent level. Robust standard errors are in parentheses.

				Table 7. Ex	ploring the l	Interest Rate (Channel					
						Ordered Pr	obit Model					
		Estimated	coefficients		1	Marginal effe	cts of inflatic	on expectatic	ons on predict	ted probabilit	ty to invest	
Selected regressors	(1)	(2)	(3)	(4)	lower (5)	higher (6)	lower (7)	higher (8)	lower (9)	higher (10)	lower (11)	higher (12)
					Panel A . Usir	ng 12-month aht	sad inflation ex	spectations				
Real interest rate Nominal interest rate	-0.037*** (0.013)	-0.034**			0.009*** (0.003)	-0.011^{***} (0.004)	0.008**	-0.010**				
Inflation expectation		0.084* 0.084*	0.078*	0.114***				0.004) 0.030* 0.030*	-0.024*	0.027*	-0.033***	0.038***
Score			(0.043) -0.055*** (0.013)	(0.008) -0.039*** (0.008)			(+10.0)		(0.012) 0.016^{**} (0.004)	(0.014) -0.018*** (0.004)	(0.002) 0.011*** (0.002)	(0.003) -0.013*** (0.003)
Obs. Pseudo R ²	3,137 0.036	3,137 0.036	3,080 0.096	8,946 0.080	3,137 0.036		3,137 0.036		3,080 0.096		8,946 0.080	
					Panel B. Usi	ing 6-month ahe	ad inflation ex]	pectations				
Real interest rate	-0.037*** (0.013)				0.009^{***} (0.003)	-0.012*** (0.004)						
Nominal interest rate		-0.033** (0.013)					0.008** (0.003)	-0.010** (0.004)				
Inflation expectation		0.116^{**} (0.058)	0.108* (0.059)	0.135^{***} (0.033)			-0.031* (0.016)	0.038** (0.020)	-0.029* (0.016)	0.035* (0.019)	-0.038*** (0.009)	0.045^{***} (0.011)
Score			-0.071*** (0.016)	-0.038*** (0.008)					0.019*** (0.005)	-0.023*** (0.005)	0.011^{***} (0.002)	-0.012*** (0.003)
Obs.	3,137	3,137	3,080	8,946	3,137		3,137		3,080		8,946	
Pseudo R ²	0.036	0.037	0.039	0.080	0.036		0.037		0.039		0.080	
<i>Notes.</i> In this table selected explanatory specification (2); in using specification (4 qualitative explanato) are available upon rec	we report the variables (re columns (2),). The margin y variables. [uest). ***, ***	e estimated e eal interest (7) and (8) nal effects au The margin: *, * denote s	coefficients : rate, nomina the results a re computed al effects of tatistical sign	and the marg l interest rature obtained tusing the sam the other continguation of the other continguation of the same of the same of the same same of the same same same same same same same sam	inal effects e, inflation ising specifi nple mean vi itrols are re 5 and 10 pei	on the predi expectation a ication (3); ir alues for the ported in Ap rcent level. R	cted probabi and score). 1 columns (5 continuous e pendix Table obust standau	lity of repoid to columns 10, (4), (9), (3), (4), (9), (5), (5), (5), (5), (5), (5), (5), (2), (2), (2), (2), (2), (2), (2), (2	rting lower (1), (5) and (1), (5) and (10), (11), (1), (1), (1), (1), (1), (1), (or higher inv [(6) the rest [2) and (13) (the sample r ed coefficien es.	estment expedults are obtained the results ar model category are other the results are other are solved to the other oth	anditure of ined using e obtained ries for the er controls

	Using 12-month ahead inflation	Using 6-month ahead inflation
Selected regressors	expectations	expectations
	(1)	(2)
	Panel A.	Cash channel
Inflation expectation	0.105***	0.133***
_	(0.036)	(0.041)
Inflation expectation × Cash ratio	0.117	0.152*
	(0.078)	(0.082)
Score	-0.051***	-0.051***
	(0.011)	(0.011)
Score \times Cash ratio	0.071**	0.072**
	(0.035)	(0.035)
Cash ratio	-0.261	-0.282
	(0.421)	(0.419)
Obs.	6,895	6,895
Pseudo R ²	0.061	0.062

Table 8. Exploring the Cash and Debt Channel: Estimated Coefficients

	Panel B.	Debt channel
Inflation expectation	-0.015	0.061
	(0.066)	(0.069)
Inflation expectation \times Debt ratio	0.300***	0.208*
	(0.107)	(0.109)
Score	-0.046**	-0.047**
	(0.020)	(0.020)
Score \times Debt ratio	0.008	0.010
	(0.040)	(0.040)
Debt ratio	-0.407	-0.359
	(0.531)	(0.531)
Obs.	6,835	6,835
Pseudo R ²	0.066	0.066

Notes. In this table we report the estimated coefficients of selected explanatory variables using specification (5). The estimated coefficients for the other controls are not reported but available upon request. ***, **, * denote statistical significance at 1, 5 and 10 percent level. Robust standard errors are in parentheses.

			Table	9. Exploring	the Cash Cha	annel				
			Margınal e (ii	ttects of inflation square bracke	on expectation its the evaluation	is on predicted on levels for t	1 probability to he cash ratio)	o invest		
			Lower					Higher		
	[%0]	[2%]	[2%]	[25%]	[%09]	[%0]	[2%]	[2%]	[25%]	[%09]
	(1)	(2)	(3)	(4)	(5)	(9)	(L)	(8)	(6)	(10)
		Pai	nel A. Using 12	2-month ahead	inflation expec	ctations (Obs.	6,895 and Pse	endo R ² =0.06	()	
Inflation Expectation	-0.028***	-0.028***	-0.029***	-0.034***	-0.043***	0.033^{***}	0.033^{***}	0.035^{***}	0.042^{***}	0.056^{***}
	(0.011)	(0.011)	(0.010)	(0.010)	(0.013)	(0.010)	(0.010)	(0.010)	(0.010)	(0.014)
Score	0.014^{***}	0.013^{***}	0.013^{***}	0.009***	0.003	-0.016^{***}	-0.016^{***}	-0.015^{***}	-0.011^{***}	-0.003
	(0.003)	(0.003)	(0.003)	(0.003)	(0.004)	(0.004)	(0.004)	(0.003)	(0.003)	(0.006)
Cash Ratio	-0.059**	-0.058**	-0.056**	-0.048**	-0.031	0.046^{*}	0.046^{*}	0.048^{*}	0.054^{**}	0.063^{**}
	(0.025)	(0.024)	(0.024)	(0.023)	(0.020)	(0.025)	(0.025)	(0.026)	(0.027)	(0.028)
		P_{3}	unel B. Using 6	-month ahead i	nflation expect	tations (Obs. 6	5,895 and Psei	udo R ² =0.062		
Inflation Expectation	-0.037***	-0.037***	-0.039***	-0.045***	-0.054***	0.044^{***}	0.044^{***}	0.047^{***}	0.057^{***}	0.077^{***}
	(0.014)	(0.014)	(0.013)	(0.014)	(0.016)	(0.012)	(0.012)	(0.012)	(0.012)	(0.017)
Score	0.014^{***}	0.014^{***}	0.013^{***}	0.009^{***}	0.002	-0.017^{***}	-0.017***	-0.016***	-0.012***	-0.003
	(0.003)	(0.003)	(0.003)	(0.003)	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)	(0.006)
Cash Ratio	-0.080**	-0.080**	-0.077**	-0.064**	-0.041^{*}	0.080^{**}	0.081^{**}	0.084^{**}	0.093^{**}	0.103^{***}
	(0.032)	(0.032)	(0.032)	(0.028)	(0.023)	(0.034)	(0.034)	(0.034)	(0.037)	(0.038)
<i>Notes.</i> In this table we reporting lower or hig sample mean values for controls are not reportionation to the parentheses.	e report the m her investmen or the continu- ted but avail.	arginal of effe the expenditure u ous explanator able upon requ	cts selected explained explored explored expecificat y variables and lest. ***, **,	planatory varial ion (5). The ma the sample mo * denote statist	oles (inflation arginal effects odel categories tical significar	expectation, s are computed s for the quali nce at 1, 5 au	core and cash 1 for different tative explana nd 10 percent	r ratio) on the values of the ttory variable t level. Robu	e predicted pr cash ratio an s. Estimates f st standard e	obability of ad using the or the other trors are in

		T	able 10. Explc	oring the Char.	mels: Role of	the Debt Ra.	tio			
			Marginal e	ffects of inflati	on expectation	s on predicted	l probability to	o invest		
			(i)	n square bracke	ts the evaluation	on levels for the	ne debt ratio)			
			Lower					Higher		
	[20%]	[30%]	[20%]	[%0]	[%08]	[20%]	[30%]	[20%]	[%02]	[%08]
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)
		Pa	nel A. Using 12	2-month ahead	inflation expec	tations (Obs.	6,835 and Pse	udo R ² =0.066	()	
Inflation Expectation	-0.013	-0.023*	-0.036***	-0.052***	-0.061***	0.015	0.028^{**}	0.044^{***}	0.063^{***}	0.074^{***}
	(0.014)	(0.012)	(0.011)	(0.013)	(0.015)	(0.016)	(0.013)	(0.010)	(0.011)	(0.013)
Score	0.012^{***}	0.012^{***}	0.012^{***}	0.011^{***}	0.011^{**}	-0.015***	-0.015***	-0.014***	-0.014***	-0.013**
	(0.004)	(0.003)	(0.003)	(0.004)	(0.005)	(0.005)	(0.004)	(0.003)	(0.005)	(0.006)
Debt Ratio	-0.012	-0.007	-0.000	0.008	0.013	-0.016	-0.012	-0.006	0.001	0.006
	(0.033)	(0.033)	(0.033)	(0.033)	(0.033)	(0.039)	(0:039)	(0.039)	(0.039)	(0.039)
		$\mathbf{P}_{\mathbf{a}}$	inel B . Using 6	-month ahead in	nflation expect	tations (Obs. 6	6,835 and Pseu	ido R ² =0.066		
Inflation Expectation	-0.028*	-0.035**	-0.044***	-0.055***	-0.062***	0.034^{**}	0.043^{***}	0.054^{***}	0.067^{***}	0.075***
	(0.015)	(0.014)	(0.013)	(0.015)	(0.017)	(0.017)	(0.014)	(0.011)	(0.012)	(0.014)
Score	0.012^{***}	0.012^{***}	0.011^{***}	0.011^{***}	0.011^{**}	-0.015***	-0.015***	-0.014***	-0.013***	-0.013**
	(0.004)	(0.003)	(0.003)	(0.004)	(0.005)	(0.005)	(0.004)	(0.003)	(0.005)	(0.006)
Debt Ratio	-0.009	-0.004	0.002	0.010	0.015	-0.018	-0.014	-00.00	-0.002	0.002
	(0.033)	(0.033)	(0.033)	(0.033)	(0.033)	(0.039)	(0.039)	(0.039)	(0.039)	(0.039)
<i>Notes.</i> In this table we reporting lower or hig sample mean values fo controls are not repor parentheses.	treport the m her investmen or the continue ted but avails	arginal of effe t expenditure t ous explanator able upon requ	cts selected exj using specificat y variables and test. ***, **,	planatory varial ion (5). The m the sample mo * denote statist	bles (inflation arginal effects del categories tical significan	expectation, s are computed for the qualition for at 1, 5 ar	core and debt I for different tative explana id 10 percent	t ratio) on the values of the tory variables level. Robu	predicted pr debt ratio ar s. Estimates f st standard er	obability of d using the or the other rors are in

ONLINE APPENDIX

FIRM INFLATION EXPECTATIONS

AND INVESTMENT PLANS

Adriana Grasso

Tiziano Ropele

LUISS Guido Carli & Einaudi Institute for Economics and Finance Bank of Italy



Appendix Figure 1. Mean Values and Standard Deviation of Selected Variables: Developments Over Time.

Notes. In this table we report the time developments of the cross-sectional mean values and standard deviation of selected variables. Statistics are calculated using sampling weights and are expressed in percentages. In top panels, we report statistics for the nominal interest rates on new term loans calculated using the original data as well as the adjusted data. "Adjusted" interest rates refer to the interest rates obtained with the imputation method for missing values described in Section 2.3. In middle panels, we report statistics for the (ex-ante) real interest rates, calculated as the difference between the nominal interest rate and inflation expectations. In bottom panels, we report statistics for the cash ratio (the sum of cash and marketable securities over current liabilities, i.e. debts due within one year) and the debt ratio (amount of current liabilities over total assets).

Row	Qualitative variables	Mean values category o	s of <i>qualitative variable</i> f expected investment	s by response expenditure
		lower	about the same	higher
(1)	Own-product expected demand	-0.13	0.10	0.25
(2)	Expected employment	-0.29	-0.09	0.12
(3)	Labor cost for expected price change	0.28	0.31	0.33
(4)	Price of raw materials for expected price change	0.29	0.31	0.32
(5)	Current conditions to invest	-0.32	-0.07	0.09
(6)	Access conditions to credit	-0.26	-0.06	0.05
(7)	Italy's macroeconomic outlook	-0.36	-0.11	0.01
(8)	Probability of improvement of Italy's outlook	-0.85	-0.82	-0.72
Row	Quantitative variables	Mean values category o	of <i>quantitative</i> variable f expected investment	es by response expenditure
		lower	about the same	higher
(9)	Number of employees (in log)	4.70	4.64	4.84
(10)	Own-price expected growth rate	-0.21	0.37	1.03
(11)	Inflation expectation (12m)	-0.04	-0.04	0.03
(12)	Inflation expectation (6m)	-0.03	-0.03	0.03
	Interest rate on new term loans:			
(13)	Nominal	0.50	0.31	-0.09
(14)	Ex-ante real (12m)	0.51	0.27	-0.18
(15)	Ex-ante real (6m)	0.50	0.25	-0.18
(16)	Cash ratio	18.52	18.90	21.92
(17)	Debt ratio	51.87	50.20	48.51

Notes. In this table we report the mean values of selected qualitative and quantitative variables distinguishing by the response categories to the question that asks firms whether they expect their investment expenditure to be "lower", "about the same" or "higher". Statistics are computed on pooled data over the period 2012Q4 to 2016Q4 using sampling weights. The 3-category *qualitative* variables reported in rows (1), (2), (5), (6) and (7) (see Table 1 in the paper) are coded as "-1", "0", "+1". The 7-category *qualitative* variables reported in rows (3) and (4) are coded as "-3", "-2", "-1", "0", "+1", "+2", "+3". Finally, the 5-category qualitative variable reported in rows (8) is coded as "-2", "-1", "0", "+1", "+2". For the 12-month and 6-month ahead inflation expectations (rows (11) and (12)) and the interest rates on new term loans (rows (13)-(15)) the mean values are computed on year-quarter demeaned data. The cash ratio is calculated as the sum of cash and marketable securities over current liabilities (i.e. debts due within one year) while the debt ratio is the amount of current liabilities over total assets.

		Ordered probit e	stimation	
	Marginal effects	on predicted probab	oility to invest	Coefficient
Selected regressors	lower	about the same	higher	
Science regressors	(1)	(2)	(3)	(4)
Inflation expectations 12 months ahead	-0.033***	-0.006**	0.038***	0.116***
	(0.008)	(0.003)	(0.009)	(0.028)
Expected labor demand in next 3 months (omitted	category "unchang	ged")		
Lower	0.116***	-0.010	-0.106***	-0.362***
	(0.013)	(0.007)	(0.010)	(0.037)
Higher	-0.092***	-0.049***	0.141***	0.390***
	(0.010)	(0.011)	(0.019)	(0.048)
Expected total demand in next 3 months (omitted of	category "unchang	eď")		
Lower	0.071***	-0.000	-0.071***	-0.232***
	(0.016)	(0.005)	(0.014)	(0.049)
Higher	-0.039***	-0.012**	0.050***	0.147***
	(0.009)	(0.005)	(0.013)	(0.036)
Expected price change in next 12 months	-0.003***	-0.001**	0.003***	0.010***
	(0.001)	(0.000)	(0.001)	(0.003)
Factors affecting expected price change in next 12	months:			
Cost of labor	-0.011***	-0.002*	0.013***	0.039***
	(0.004)	(0.001)	(0.005)	(0.015)
Price of raw materials	0.002	0.000	-0.003	-0.008
	(0.004)	(0.001)	(0.005)	(0.014)
Current conditions to invest compared with previou	us 3 months (omit	ted category "unchai	nged")	
Worse	0.091***	-0.004	-0.087***	-0.290***
	(0.016)	(0.006)	(0.013)	(0.047)
Better	-0.084***	-0.042***	0.127***	0.352***
	(0.011)	(0.011)	(0.020)	(0.052)
Current access conditions to credit compared with	previous 3 months	(omitted category "	unchanged")	
Worse	0.079***	-0.001	-0.078***	-0.256***
	(0.015)	(0.006)	(0.013)	(0.045)
Better	-0.060***	-0.023***	0.083***	0.237***
	(0.013)	(0.009)	(0.020)	(0.054)
Number of employees (in natural logarithm)	-0.003	-0.001	0.004	0.012
	(0.004)	(0.001)	(0.004)	(0.013)
Italy's current economic outlook compared with pr	revious 3 months (omitted category "u	nchanged")	
Worse	0.015	0.002	-0.017	-0.051
	(0.012)	(0.002)	(0.014)	(0.042)
Better	-0.022*	-0.005	0.027	0.081

A	ppendix Table 2	. Effects o	of 12-Month	Ahead Inflat	ion Expectati	ions on Expe	ected Investment	Expenditure
			,	· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·	- ··· · · · · · · · · · · · · · · · · ·		··· · · · · · · · · · · · · · · · · ·

continues from previous page

Probability of an improvement in Italy's g	eneral economic	situation in next 3	months (omitted cate	gory "Zero"):
1-25 percent	-0.011	-0.002	0.013	0.039
	(0.009)	(0.002)	(0.011)	(0.032)
26-50 percent	-0.011	-0.002	0.013	0.040
	(0.015)	(0.003)	(0.018)	(0.053)
51-75 percent	0.015	0.002	-0.017	-0.052
	(0.022)	(0.002)	(0.023)	(0.074)
75-99 percent	0.013	0.001	-0.014	-0.044
	(0.063)	(0.005)	(0.068)	(0.213)
Share of revenues from exports (omitted ca	ategory "0 perce	<i>nt</i> "):		
0-33 percent	-0.011	-0.002	0.014	0.041
	(0.011)	(0.002)	(0.013)	(0.038)
33-66 percent	0.010	0.001	-0.011	-0.033
	(0.013)	(0.002)	(0.015)	(0.045)
66 percent and over	-0.033**	-0.009**	0.042**	0.123**
	(0.013)	(0.004)	(0.017)	(0.049)
Geographical area (omitted category "North	th-West")			
North-East	-0.003	-0.000	0.003	0.010
	(0.010)	(0.002)	(0.012)	(0.036)
Centre	0.010	0.002	-0.012	-0.036
	(0.011)	(0.002)	(0.013)	(0.038)
South and Islands	-0.018*	-0.004	0.022*	0.066*
	(0.010)	(0.003)	(0.013)	(0.038)
Observations $= 9615$				
Pseudo R-square = 0.083				

Notes. In this table we report the baseline estimates of specification (1) to study the effects of firms' 12-month ahead inflation expectations on their expected investment expenditure. We report the estimated marginal effects (columns (1)-(3)) as well as the estimated coefficient (column (4)) of the main regressors. The marginal effects are computed using the full-sample mean values for the continuous explanatory variables and the full-sample mode categories for the qualitative explanatory variables. Estimates of sector and time fixed effects are not reported. ***, **, * denote statistical significance at 1, 5 and 10 percent level. Robust standard errors are in parentheses.

		Ordered probit es	stimation	
	Marginal effects	Coefficient		
Degreggers	lower	about the same	higher	
Regressors	(1)	(2)	(3)	(4)
	0.027***	0.00/**	0.042***	0 101444
Inflation expectations 6 months ahead	-0.037***	-0.006**	0.043***	0.131***
	(0.009)	(0.003)	(0.011)	(0.032)
Expected labor demand in next 3 months (omitted	category "unchang	ged")		
Lower	0.115***	-0.010	-0.105***	-0.361***
	(0.013)	(0.007)	(0.010)	(0.037)
Higher	-0.092***	-0.049***	0.140***	0.388***
	(0.010)	(0.011)	(0.019)	(0.048)
Expected total demand in next 3 months (omitted of	category "unchange	ed")		
Lower	0.071***	-0.000	-0.071***	-0.233***
	(0.016)	(0.005)	(0.014)	(0.049)
Higher	-0.039***	-0.012**	0.050***	0.147***
	(0.009)	(0.005)	(0.013)	(0.036)
Expected price change in next 12 months	-0.003***	-0.001**	0.003***	0.010***
1 1 0	(0.001)	(0.000)	(0.001)	(0.003)
Factors affecting expected price change in next 12	months			
Cost of labor	-0.011**	-0.002*	0.013**	0.039**
	(0.004)	(0.002)	(0.015)	(0.03)
Price of raw materials	0.002	0.000	-0.002	-0.007
	(0.004)	(0.001)	(0.005)	(0.014)
Current conditions to invest compared with previo	us 3 months (omitt	ed category "uncha	naad")	
Worse	0 000***		-0 087***	-0 289***
worse	(0.016)	(0.004)	(0.007)	(0.047)
Better	-0.084***	-0.042***	0.126***	0.351***
	(0.011)	(0.011)	(0.020)	(0.052)
Comment access and divisions to an dit according durith		(amittad astacom "	······································	× ,
Worse			0 078***	0.257***
worse	(0.015)	(0.002)	(0.013)	(0.045)
Rottor	-0.060***	-0.023***	0.084***	0.238***
Dener	(0.013)	(0.023)	(0.004)	(0.055)
Number of employees (in natural logarithm)	-0.004	-0.001	0.004	0.013
rumber of employees (in natural logarithm)	(0.004)	(0.001)	(0.004)	(0.012)
Italy's aument accurate authority and with m	errieure 2 monthe (itted actorsmy "	(
Worke	1 evious 3 months (0)	o ooo	ncnangea)	0.051
worse	(0.013)	(0.002)	-0.017	-0.031
Bottor	(0.012)	-0.002)	(0.014) 0.027	(0.042)
Denet	(0.013)	(0.003)	(0.027)	(0.049)
	(0.015)	(0.00+)	(0.017)	(0.0+)

Probabilit	v of an i	mprovement	in Italy'	s genera	l economic	situation	in next 3	months	omitted	category '	"Zero"):
1100401110	<i>j</i> • • • • • • • •			Benera	•••••	010000001011		1110110110	(01110000	• me Berj		<i>,.</i>

1-25 percent	-0.012	-0.002	0.014	0.042
1-25 percent	(0.009)	(0.002)	(0.014)	(0.042)
26-50 percent	-0.013	-0.002	0.016	0.047
20-50 percent	(0.015)	(0.002)	(0.018)	(0.053)
51-75 percent	(0.013)	0.003)	-0.013	-0.040
51-75 percent	(0.012)	(0.001)	(0.023)	(0.074)
$75_{-}00$ percent	(0.022)	0.002)	(0.023)	(0.074)
75-77 percent	(0.063)	(0.005)	(0.068)	(0.213)
Share of revenues from exports (omitted cate	egory "0 perc	ent"):		
0-33 percent	-0.011	-0.002	0.014	0.041
I I I I I I I I I I I I I I I I I I I	(0.011)	(0.002)	(0.013)	(0.038)
33-66 percent	0.008	0.001	-0.009	-0.029
	(0.013)	(0.002)	(0.015)	(0.045)
66 percent and over	-0.033**	-0.009**	0.042**	0.124**
	(0.013)	(0.004)	(0.017)	(0.049)
Geographical area (omitted category "North-	-West")			
North-East	-0.003	-0.001	0.004	0.012
	(0.010)	(0.002)	(0.012)	(0.036)
Centre	0.011	0.002	-0.012	-0.038
	(0.011)	(0.002)	(0.013)	(0.038)
South and Islands	-0.018*	-0.004	0.022*	0.065*
	(0.010)	(0.003)	(0.013)	(0.038)
Observations = 9,615				
December 2,0002				

Notes. In this table we report the baseline estimates of specification (1) to study the effects of firms' 6-month ahead inflation expectations on their expected investment expenditure. We report the estimated marginal effects (columns (1)-(3)) as well as the estimated coefficient (column (4)) of the main regressors. The marginal effects are computed using the full-sample mean values for the continuous explanatory variables and the full-sample mode categories for the qualitative explanatory variables. Estimates of sector and time fixed effects are not reported. ***, **, * denote statistical significance at 1, 5 and 10 percent level. Robust standard errors are in parentheses.

Evaluation point for	Marginal effect of 12- expectation of probability	month ahead inflation on predicted / to invest	Marginal effect of 6-month ahead inflation expectation on predicted probability to invest			
inflation expectation	Lower	Higher	Lower	Higher		
	(1)	(2)	(3)	(4)		
-0.2% (≅ p5)	-0.034***	0.034***	-0.039***	0.039***		
	(0.009)	(0.008)	(0.010)	(0.009)		
0.0%	-0.034***	0.035***	-0.038***	0.040***		
	(0.009)	(0.008)	(0.010)	(0.009)		
0.2%	-0.033***	0.035***	-0.038***	0.040***		
	(0.009)	(0.008)	(0.010)	(0.010)		
0.4%	-0.033***	0.036***	-0.037***	0.041***		
	(0.008)	(0.009)	(0.009)	(0.010)		
0.6%	-0.032***	0.036***	-0.036***	0.042***		
	(0.008)	(0.009)	(0.009)	(0.010)		
0.8%	-0.031***	0.037***	-0.035***	0.042***		
	(0.008)	(0.009)	(0.009)	(0.011)		
1.0%	-0.031***	0.037***	-0.035***	0.043***		
	(0.008)	(0.009)	(0.008)	(0.011)		
1.2%	-0.030***	0.038***	-0.034***	0.044***		
	(0.007)	(0.010)	(0.008)	(0.011)		
1.4%	-0.030***	0.038***	-0.033***	0.044***		
	(0.007)	(0.010)	(0.008)	(0.012)		
1.6%	-0.029***	0.039***	-0.033***	0.045***		
	(0.007)	(0.010)	(0.007)	(0.012)		
1.8%	-0.029***	0.039***	-0.032***	0.045***		
	(0.006)	(0.010)	(0.007)	(0.012)		
2.0%	-0.028***	0.039***	-0.031***	0.046***		
	(0.006)	(0.010)	(0.007)	(0.012)		
2.2%	-0.027***	0.040***	-0.030***	0.046***		
	(0.006)	(0.010)	(0.006)	(0.012)		
2.4%	-0.027***	0.040***	-0.029***	0.047***		
	(0.006)	(0.011)	(0.006)	(0.013)		
2.6%	-0.026***	0.040***	-0.029***	0.047***		
	(0.005)	(0.011)	(0.005)	(0.013)		
2.8% (≅ p95)	-0.026***	0.041***	-0.028***	0.047***		
	(0.005)	(0.011)	(0.005)	(0.013)		

Appendix Table 4. Marginal Effects of Inflation Expectations on Business Investment Plan
Evaluated At Different Values of Inflation Expectations

Notes. In this table we report the marginal effects of firms' inflation expectations on the predicted probability of reporting "lower" or "higher" expected investment evaluated. The marginal effects are obtained using the baseline specification and evaluated for different values of inflation expectations in the range from -0.2 to 2.8 percent, corresponding approximately to the 1st and 99th percentile of the empirical distribution of inflation expectations. The other regressors are evaluated at benchmark values as in Tables A2 and A3. **, and *** denote statistical significance at the 5% and 1% critical level, respectively. Robust standard errors are in parenthesis.

			Ordered pro	bit estimation		
	Marginal effects on predicted probability to invest					
	much	A little	About the	A little	Much	coefficient
Regressors	lower	lower	same	higher	higher	
	(1)	(2)	(3)	(4)	(5)	(6)
Inflation expectations 6 months ahead	-0.013***	-0.019***	-0.007**	0.030***	0.008***	0.114***
	(0.003)	(0.005)	(0.003)	(0.008)	(0.002)	(0.030)
Expected labor demand in next 3 months (omitted catego	orv "unchange	d'')			
Lower	0.054***	0.063***	-0.007	-0.091***	-0.019***	-0.372***
	(0.007)	(0.006)	(0.007)	(0.009)	(0.002)	(0.036)
Higher	-0.030***	-0.054***	-0.045***	0.094***	0.034***	0.355***
	(0.004)	(0.006)	(0.010)	(0.012)	(0.006)	(0.045)
Expected total demand in next 3 months (o	mitted catego	ry " <i>unchanged</i>	<i>l</i> ")			
Lower	0.028***	0.038***	0.002	-0.056***	-0.012***	-0.218***
	(0.007)	(0.008)	(0.004)	(0.012)	(0.002)	(0.048)
Higher	-0.014***	-0.024***	-0.013***	0.039***	0.011***	0.146***
	(0.003)	(0.005)	(0.004)	(0.009)	(0.003)	(0.033)
Expected price change in next 12 months	-0.001***	-0.002***	-0.001***	0.003***	0.001***	0.010***
I I I I I I I I I I I I I I I I I I I	(0.000)	(0.000)	(0.000)	(0.001)	(0.000)	(0.003)
Factors affecting expected price change in	nevt 12 mont	.		. ,	, ,	
Cost of labor	_0 005***	_0 007***	-0 003**	0.011***	0.003***	0 0/3***
	(0.003)	(0.007)	(0.003)	(0.004)	(0.003)	(0.043)
Price of raw materials	0.001	(0.002)	0.001	-0.003	-0.001	-0.011
The of fuw materials	(0.002)	(0.002)	(0.001)	(0.003)	(0.001)	(0.014)
	(0:002)	(0.002)	1 ("		(0.001)	(0.011)
Current conditions to invest compared with	n previous 3 m	ionths (omittee	a category "un	changed)	0 01 4***	0 251***
worse	0.034^{***}	0.043***	(0.001)	-0.063***	-0.014***	-0.251***
D - 44	(0.007)	(0.008)	(0.005)	(0.011)	(0.002)	(0.040)
Бенег	-0.028^{+++}	-0.049^{++++}	-0.039^{+++}	$(0.080^{-1.12})$	(0.030^{-101})	(0.046)
	(0.004)	(0.007)	(0.009)	(0.012)	(0.000)	(0.040)
Current access conditions to credit compar-	ed with previo	ous 3 months (omitted catego	ory "unchange	<i>d</i> ")	
Worse	0.043***	0.053***	-0.002	-0.077***	-0.016***	-0.311***
_	(0.008)	(0.008)	(0.006)	(0.010)	(0.002)	(0.044)
Better	-0.021***	-0.036***	-0.023***	0.060***	0.019***	0.226***
	(0.004)	(0.007)	(0.008)	(0.013)	(0.005)	(0.050)
Number of employees (in natural log)	-0.003**	-0.004**	-0.001*	0.007**	0.002**	0.026**
	(0.001)	(0.002)	(0.001)	(0.003)	(0.001)	(0.012)
Italy's current economic outlook compared	with previou	s 3 months (or	nitted category	("unchanged"	")	
Worse	0.006	0.009	0.003	-0.015	-0.004	-0.056
	(0.005)	(0.007)	(0.002)	(0.011)	(0.003)	(0.042)
Better	-0.007*	-0.011	-0.005	0.018	0.005	0.069
	(0.004)	(0.007)	(0.004)	(0.012)	(0.003)	(0.043)

Appendix Table 5. Robustness: Effects of 6-Month Ahead Inflation Expectations on Expected Investment Expenditure (Coded As 5-Category Dependent Variable)

				(Continues from	i previous po
Probability of an improvement in	Italy's general econom	nic situation in	next 3 months	(omitted cate	gory "Zero"):	
1-25 percent	-0.004	-0.006	-0.002	0.010	0.003	0.038
-	(0.003)	(0.005)	(0.002)	(0.008)	(0.002)	(0.030)
26-50 percent	-0.007	-0.010	-0.004	0.016	0.004	0.060
-	(0.005)	(0.008)	(0.003)	(0.013)	(0.004)	(0.049)
51-75 percent	0.001	0.002	0.000	-0.002	-0.001	-0.009
-	(0.008)	(0.011)	(0.003)	(0.018)	(0.004)	(0.067)
75-99 percent	-0.001	-0.001	-0.000	0.002	0.000	0.007
-	(0.019)	(0.028)	(0.009)	(0.045)	(0.012)	(0.171)
Share of revenues from exports (c	mitted category "0 per	cent")				
0-33 percent	-0.008**	-0.012**	-0.005*	0.020**	0.006**	0.075**
	(0.004)	(0.006)	(0.003)	(0.009)	(0.003)	(0.036)
33-66 percent	0.003	0.005	0.001	-0.007	-0.002	-0.028
	(0.005)	(0.007)	(0.002)	(0.011)	(0.003)	(0.042)
66 percent and over	-0.011**	-0.017**	-0.008**	0.028**	0.008**	0.106**
	(0.005)	(0.007)	(0.004)	(0.012)	(0.004)	(0.046)
Geographical area (omitted catego	ory "North-West")					
North-East	-0.000	-0.001	-0.000	0.001	0.000	0.003
	(0.004)	(0.006)	(0.002)	(0.009)	(0.002)	(0.034)
Centre	0.004	0.005	0.002	-0.008	-0.002	-0.031
	(0.004)	(0.006)	(0.002)	(0.010)	(0.003)	(0.037)
South and Islands	-0.004	-0.006	-0.002	0.009	0.003	0.035

Observations = 9.615

Pseudo R-square = 0.070

Notes. In this table we report the baseline estimates of specification (1) to study the effects of firms' 6-month ahead inflation expectations on their expected investment expenditure. We report the estimated marginal effects (columns (1)-(3)) as well as the estimated coefficient (column (4)) of the main regressors. The marginal effects are computed using the full-sample mean values for the continuous explanatory variables and the full-sample mode categories for the qualitative explanatory variables. Estimates of sector and time fixed effects are not reported (available upon request). ***, **, * denote statistical significance at 1, 5 and 10 percent level. Robust standard errors are in parentheses.

(0.006)

(0.002)

(0.009)

(0.003)

(0.035)

(0.004)

			Ordered pro	bit estimation		
	Mai	ginal effects o	on predicted pro	obability to in	vest	Coefficient
	much	A little	About the	A little	Much	
Regressors	lower	lower	same	higher	higher	
	(1)	(2)	(3)	(4)	(5)	(6)
Inflation expectations 12 months ahead	-0.010***	-0.016***	-0.006**	0.025***	0.007***	0.095***
-	(0.003)	(0.004)	(0.002)	(0.007)	(0.002)	(0.026)
Expected labor demand in next 3 months (c	mitted catego	ory "unchange	eď")			
Lower	0.054***	0.063***	-0.008	-0.091***	-0.019***	-0.372***
	(0.007)	(0.006)	(0.007)	(0.009)	(0.002)	(0.036)
Higher	-0.030***	-0.054***	-0.045***	0.095***	0.034***	0.356***
	(0.004)	(0.006)	(0.010)	(0.012)	(0.006)	(0.045)
Expected total demand in next 3 months (or	mitted catego	ry " <i>unchanged</i>	<i>d</i> ")			
Lower	0.028***	0.037***	0.002	-0.056***	-0.012***	-0.217***
	(0.007)	(0.008)	(0.004)	(0.012)	(0.002)	(0.048)
Higher	-0.014***	-0.024***	-0.013***	0.039***	0.011***	0.146***
	(0.003)	(0.005)	(0.004)	(0.009)	(0.003)	(0.033)
Expected price change in next 12 months	-0.001***	-0.002***	-0.001***	0.003***	0.001***	0.010***
	(0.000)	(0.000)	(0.000)	(0.001)	(0.000)	(0.003)
Factors affecting expected price change in a	next 12 montl	ns:				
Cost of labor	-0.005***	-0.007***	-0.003**	0.012***	0.003***	0.044***
	(0.002)	(0.002)	(0.001)	(0.004)	(0.001)	(0.014)
Price of raw materials	0.001	0.002	0.001	-0.003	-0.001	-0.012
	(0.002)	(0.002)	(0.001)	(0.004)	(0.001)	(0.014)
Current conditions to invest compared with	previous 3 m	onths (omitte	d category "un	changed")		
Worse	0.034***	0.043***	0.001	-0.064***	-0.014***	-0.252***
	(0.007)	(0.008)	(0.005)	(0.011)	(0.002)	(0.046)
Better	-0.028***	-0.050***	-0.039***	0.086***	0.030***	0.324***
	(0.004)	(0.007)	(0.010)	(0.012)	(0.006)	(0.046)
Current access conditions to credit compare	ed with previo	ous 3 months (omitted catego	rv "unchange	<i>d</i> ")	
Worse	0.043***	0.053***	-0.002	-0.077***	-0.016***	-0.310***
	(0.008)	(0.008)	(0.006)	(0.010)	(0.002)	(0.044)
Better	-0.021***	-0.035***	-0.023***	0.060***	0.019***	0.225***
	(0.004)	(0.008)	(0.008)	(0.013)	(0.005)	(0.050)
Number of employees (in natural log)	-0.003**	-0 004**	-0.001*	0.007**	0.002**	0.025**
ivaluation of employees (in natural log)	(0.003)	(0.004)	(0.001)	(0.007)	(0.002)	(0.023)
T 1 2 / · · · 1 1 1	(0.001)	(0.002)	(0.001)	(0.005)	(0.001)	(0.012)
Italy's current economic outlook compared	with previou	s 3 months (or	mitted category	""unchanged"	⁽)	0.056
Worse	0.006	0.009	0.003	-0.015	-0.004	-0.056
	(0.005)	(0.007)	(0.002)	(0.011)	(0.003)	(0.042)
Better	-0.007*	-0.011	-0.005	0.018	0.005	0.070
	(0.004)	(0.007)	(0.004)	(0.012)	(0.003)	(0.043)

Appendix Table 6. Robustness: Effects of 12-Month Ahead Inflation Expectations on Expected Investment Expenditure (Coded As 5-Category Dependent Variable)

Probability of an improvement in Ita	ly's general econom	ic situation in	next 3 months	(omitted cate	gory "Zero"):	
1-25 percent	-0.004	-0.006	-0.002	0.009	0.002	0.036
-	(0.003)	(0.005)	(0.002)	(0.008)	(0.002)	(0.030)
26-50 percent	-0.006	-0.009	-0.003	0.014	0.004	0.054
-	(0.005)	(0.008)	(0.003)	(0.013)	(0.004)	(0.049)
51-75 percent	0.002	0.003	0.001	-0.005	-0.001	-0.019
	(0.008)	(0.011)	(0.003)	(0.018)	(0.004)	(0.067)
75-99 percent	-0.000	-0.000	-0.000	0.000	0.000	0.001
	(0.019)	(0.029)	(0.009)	(0.045)	(0.012)	(0.171)
Share of revenues from exports (om	itted category "0 per	cent")				
0-33 percent	-0.008**	-0.012**	-0.006*	0.020**	0.006**	0.076**
	(0.004)	(0.006)	(0.003)	(0.010)	(0.003)	(0.036)
33-66 percent	0.003	0.005	0.002	-0.008	-0.002	-0.031
	(0.005)	(0.007)	(0.002)	(0.011)	(0.003)	(0.042)
66 percent and over	-0.011**	-0.017**	-0.008**	0.028**	0.008**	0.106**
	(0.005)	(0.007)	(0.004)	(0.012)	(0.004)	(0.046)
Geographical area (omitted category	""North-West")					
North-East	-0.000	-0.000	-0.000	0.000	0.000	0.001
	(0.004)	(0.006)	(0.002)	(0.009)	(0.002)	(0.034)
Centre	0.003	0.005	0.002	-0.008	-0.002	-0.031
	(0.004)	(0.006)	(0.002)	(0.010)	(0.003)	(0.037)
South and Islands	-0.004	-0.006	-0.002	0.009	0.003	0.036
	(0.004)	(0.006)	(0.002)	(0.009)	(0.003)	(0.035)
Observations = 9,615						
Pseudo R-square $= 0.070$						

Notes. In this table we report the baseline estimates of specification (1) to study the effects of firms' 6-month ahead inflation expectations on their expected investment expenditure. We report the estimated marginal effects (columns (1)-(3)) as well as the estimated coefficient (column (4)) of the main regressors. The marginal effects are computed using the full-sample mean values for the continuous explanatory variables and the full-sample mode categories for the qualitative explanatory variables. Estimates of sector and time fixed effects are not reported (available upon request). ***, **, * denote statistical significance at 1, 5 and 10 percent level. Robust standard errors are in parentheses.

	Ordered probit estimation						
	Mar	ginal effects o	on predicted pro	obability to inv	vest	Coefficient	
	much	A little	About the	A little	Much		
Regressors	lower	lower	same	higher	higher		
	(1)	(2)	(3)	(4)	(5)	(6)	
Inflation expectations 2 years ahead	-0.008***	-0.012***	-0.004**	0.019***	0.005***	0.072***	
	(0.002)	(0.003)	(0.002)	(0.005)	(0.001)	(0.019)	
Expected labor demand in next 3 months (or	omitted catego	ory "unchange	<i>d</i> ")				
Lower	0.054***	0.063***	-0.007	-0.092***	-0.018***	-0.372***	
	(0.007)	(0.006)	(0.007)	(0.009)	(0.002)	(0.036)	
Higher	-0.029***	-0.054***	-0.045***	0.095***	0.034***	0.355***	
	(0.004)	(0.006)	(0.010)	(0.012)	(0.006)	(0.045)	
Expected total demand in next 3 months (o	mitted catego	ry " <i>unchanged</i>	<i>l</i> ")				
Lower	0.028***	0.038***	0.002	-0.056***	-0.012***	-0.217***	
	(0.007)	(0.008)	(0.004)	(0.012)	(0.002)	(0.047)	
Higher	-0.014***	-0.024***	-0.013***	0.040***	0.011***	0.147***	
	(0.003)	(0.005)	(0.004)	(0.009)	(0.003)	(0.033)	
Expected price change in next 12 months	-0.001***	-0.002***	-0.001***	0.003***	0.001***	0.010***	
Expected price change in next 12 months	(0.000)	(0.000)	(0.000)	(0.001)	(0.000)	(0.003)	
	(0.000)	(0.000)	(0.000)	(0.001)	(0.000)	(0.000)	
Factors affecting expected price change in a	next 12 month	1S:	0.002**	0 011***	0.002***	0.042***	
Cost of labor	-0.005***	-0.00/***	-0.003**	0.011^{***}	0.003^{***}	0.043^{***}	
	(0.002)	(0.002)	(0.001)	(0.004)	(0.001)	(0.014)	
Price of raw materials	0.001	0.002	0.001	-0.003	-0.001	-0.011	
	(0.002)	(0.002)	(0.001)	(0.004)	(0.001)	(0.014)	
Current conditions to invest compared with	previous 3 m	nonths (omittee	d category "une	changed")			
Worse	0.033***	0.043***	0.001	-0.064***	-0.014***	-0.251***	
	(0.007)	(0.008)	(0.005)	(0.011)	(0.002)	(0.046)	
Better	-0.027***	-0.050***	-0.039***	0.086***	0.030***	0.322***	
	(0.004)	(0.007)	(0.010)	(0.012)	(0.006)	(0.046)	
Current access conditions to credit compare	ed with previo	ous 3 months (omitted catego	rv "unchange	<i>d</i> ")		
Worse	0.044***	0.053***	-0.002	-0.078***	-0.016***	-0.313***	
	(0.008)	(0.008)	(0.006)	(0.010)	(0.002)	(0.044)	
Better	-0.021***	-0.036***	-0.023***	0.061***	0.019***	0.226***	
	(0.004)	(0.008)	(0.008)	(0.013)	(0.005)	(0.050)	
Number of employees (in natural log)	-0.003**	-0.004**	-0.001*	0.007**	0.002**	0.025**	
	(0.001)	(0.002)	(0.001)	(0.003)	(0.001)	(0.012)	
Italy's current economic outlook compared	with previou	s 3 months (or	nitted category	"unchanged")		
Worse	0.006	0.009	0.003	-0.015	-0.004	-0.056	
noise	(0.005)	(0.007)	(0.002)	(0.011)	(0.003)	(0.042)	
Better	-0.007*	-0.011	-0.005	0.019	0.005	0.070	
	(0.004)	(0.007)	(0.004)	(0.012)	(0.003)	(0.043)	

Appendix Table 7. Robustness: Effects of 2-Year Ahead Inflation Expectations on Expected Investment Expenditure (Coded As 5-Category Dependent Variable)

Probability of an improvement in I	taly's general econom	nic situation in	next 3 months	(omitted cate	gory "Zero")	
1-25 percent	-0.004	-0.006	-0.002	0.009	0.002	0.035
	(0.003)	(0.005)	(0.002)	(0.008)	(0.002)	(0.030)
26-50 percent	-0.006	-0.008	-0.003	0.013	0.004	0.051
	(0.005)	(0.008)	(0.003)	(0.013)	(0.004)	(0.049)
51-75 percent	0.003	0.004	0.001	-0.006	-0.001	-0.022
	(0.008)	(0.011)	(0.003)	(0.018)	(0.004)	(0.067)
75-99 percent	0.000	0.000	0.000	-0.000	-0.000	-0.001
	(0.019)	(0.029)	(0.009)	(0.045)	(0.011)	(0.171)
Share of revenues from exports (or	nitted category "0 per	cent")				
0-33 percent	-0.008**	-0.012**	-0.006*	0.020**	0.006**	0.074**
	(0.004)	(0.006)	(0.003)	(0.010)	(0.003)	(0.036)
33-66 percent	0.004	0.006	0.002	-0.009	-0.002	-0.035
	(0.005)	(0.007)	(0.002)	(0.011)	(0.003)	(0.042)
66 percent and over	-0.010**	-0.017**	-0.008**	0.027**	0.008**	0.102**
	(0.005)	(0.007)	(0.004)	(0.012)	(0.004)	(0.046)
Geographical area (omitted categor	ry "North-West")					
North-East	0.000	0.000	0.000	-0.000	-0.000	-0.000
	(0.004)	(0.006)	(0.002)	(0.009)	(0.002)	(0.034)
Centre	0.003	0.005	0.002	-0.008	-0.002	-0.030
	(0.004)	(0.006)	(0.002)	(0.010)	(0.003)	(0.037)
South and Islands	-0.004	-0.006	-0.003	0.010	0.003	0.039
	(0.004)	(0.006)	(0.002)	(0.009)	(0.003)	(0.035)
Observations = 9,615						
Pseudo R-square $= 0.070$						

Notes. In this table we report the baseline estimates of specification (1) to study the effects of firms' 6-month ahead inflation expectations on their expected investment expenditure. We report the estimated marginal effects (columns (1)-(3)) as well as the estimated coefficient (column (4)) of the main regressors. The marginal effects are computed using the full-sample mean values for the continuous explanatory variables and the full-sample mode categories for the qualitative explanatory variables. Estimates of sector and time are not reported (available upon request). ***, **, * denote statistical significance at 1, 5 and 10 percent level. Robust standard errors are in parentheses.

			Ordered pro	bit estimation		
	Mai	rginal effects o	on predicted pro	obability to in	vest	Coefficient
	much	A little	About the	A little	Much	
Regressors	lower	lower	same	higher	higher	
	(1)	(2)	(3)	(4)	(5)	(6)
Inflation expectations 2 years ahead from	-0.006***	-0.009***	-0.005**	0.016***	0.004***	0.058***
2 years	(0.002)	(0.003)	(0.002)	(0.005)	(0.001)	(0.018)
Expected labor demand in next 3 months (omitted catego	ory "unchange	<i>d</i> ")			
Lower	0.043***	0.054***	0.006	-0.085***	-0.017***	-0.329***
	(0.007)	(0.007)	(0.007)	(0.011)	(0.003)	(0.043)
Higher	-0.027***	-0.051***	-0.058***	0.100***	0.035***	0.367***
	(0.004)	(0.006)	(0.012)	(0.013)	(0.007)	(0.050)
Expected total demand in next 3 months (o	mitted catego	ry "unchanged	ď")			
Lower	0.032***	0.044***	0.008	-0.070***	-0.014***	-0.266***
	(0.009)	(0.010)	(0.006)	(0.015)	(0.003)	(0.060)
Higher	-0.010***	-0.016***	-0.012**	0.030***	0.008***	0.106***
	(0.003)	(0.006)	(0.005)	(0.010)	(0.003)	(0.037)
Expected price change in next 12 months	-0.001***	-0.001***	-0.001**	0.002***	0.001***	0.009***
	(0.000)	(0.000)	(0.000)	(0.001)	(0.000)	(0.003)
Factors affecting expected price change in	novt 12 mont					
Cost of labor	-0 004***	-0.007***	-0 004**	0.012***	0 003***	0 044***
	(0.004)	(0.007)	(0.004)	(0.012)	(0.003)	(0.044)
Price of raw materials	-0.001	-0.001	-0.001	0.002	0.001	0.008
Thee of fuw materials	(0.001)	(0.001)	(0.001)	(0.002)	(0.001)	(0.016)
	(0.002)	(0.002)	(0.001)		(0.001)	(0.010)
Current conditions to invest compared with	n previous 3 m	nonths (omittee	d category "un	changed")	0.017***	0 227***
Worse	0.044***	0.055***	0.005	-0.08/***	-0.01/***	-0.33/***
D - 44	(0.010)	(0.011)	(0.008)	(0.015)	(0.003)	(0.062)
Better	-0.025^{++++}	-0.04/	-0.051^{****}	0.092^{****}	0.031^{****}	0.334^{****}
	(0.004)	(0.007)	(0.012)	(0.014)	(0.000)	(0.030)
Current access conditions to credit compar	ed with previo	ous 3 months (omitted catego	ry "unchange	<i>d</i> ")	
Worse	0.030***	0.040***	0.009*	-0.065***	-0.014***	-0.248***
	(0.008)	(0.009)	(0.005)	(0.014)	(0.003)	(0.056)
Better	-0.020***	-0.035***	-0.033***	0.066***	0.021***	0.241***
	(0.004)	(0.007)	(0.011)	(0.015)	(0.006)	(0.054)
Number of employees (in natural log)	-0.003**	-0.004**	-0.002*	0.008**	0.002**	0.028**
	(0.001)	(0.002)	(0.001)	(0.004)	(0.001)	(0.013)
Italy's current economic outlook compared	d with previou	s 3 months (or	nitted category	""unchanged	')	
Worse	0.009	0.014	0.006*	-0.024*	-0.006*	-0.089
	(0.006)	(0.009)	(0.003)	(0.015)	(0.003)	(0.054)
Better	-0.005	-0.007	-0.005	0.013	0.004	0.049
	(0.004)	(0.007)	(0.005)	(0.013)	(0.004)	(0.047)

Appendix Table 8. Robustness: Effects of 2-Year Ahead From 2 Years Inflation Expectations on Expected Investment Expenditure (Coded As 5-Category Dependent Variable)

Probability of an improvement in I	taly's general econom	ic situation in	next 3 months	(omitted cate	gory "Zero")	
1-25 percent	-0.004	-0.007	-0.004	0.012	0.003	0.043
	(0.004)	(0.006)	(0.003)	(0.010)	(0.002)	(0.036)
26-50 percent	-0.007	-0.011	-0.006	0.019	0.005	0.071
-	(0.006)	(0.009)	(0.006)	(0.016)	(0.004)	(0.057)
51-75 percent	-0.003	-0.004	-0.002	0.008	0.002	0.027
	(0.008)	(0.012)	(0.006)	(0.021)	(0.005)	(0.075)
75-99 percent	-0.010	-0.015	-0.010	0.028	0.007	0.100
-	(0.016)	(0.028)	(0.021)	(0.051)	(0.014)	(0.184)
Share of revenues from exports (or	nitted category "0 per	cent")				
0-33 percent	-0.004	-0.007	-0.004	0.012	0.003	0.044
-	(0.004)	(0.006)	(0.004)	(0.011)	(0.003)	(0.042)
33-66 percent	0.009*	0.014*	0.006	-0.023*	-0.005*	-0.086*
	(0.005)	(0.008)	(0.004)	(0.013)	(0.003)	(0.049)
66 percent and over	-0.008	-0.012	-0.009	0.023	0.006	0.082
	(0.005)	(0.008)	(0.006)	(0.014)	(0.004)	(0.052)
Geographical area (omitted categor	ry "North-West")					
North-East	0.001	0.002	0.001	-0.004	-0.001	-0.014
	(0.004)	(0.006)	(0.003)	(0.011)	(0.003)	(0.039)
Centre	0.003	0.005	0.003	-0.009	-0.002	-0.033
	(0.004)	(0.007)	(0.003)	(0.012)	(0.003)	(0.043)
South and Islands	-0.003	-0.004	-0.003	0.008	0.002	0.029
	(0.004)	(0.006)	(0.004)	(0.011)	(0.003)	(0.040)
Observations = 7.332						
Pseudo R-square = 0.064						

Notes. In this table we report the baseline estimates of specification (1) to study the effects of firms' 6-month ahead inflation expectations on their expected investment expenditure. We report the estimated marginal effects (columns (1)-(3)) as well as the estimated coefficient (column (4)) of the main regressors. The marginal effects are computed using the full-sample mean values for the continuous explanatory variables and the full-sample mode categories for the qualitative explanatory variables. Estimates of sector and time fixed effects are not reported (available upon request). ***, **, * denote statistical significance at 1, 5 and 10 percent level. Robust standard errors are in parentheses.

	Mai	rginal effects	of 12-month	inflation expe	ectations on p	predicted prol	pability to inv	vest
Regressors	Lower (1)	Higher (2)	Lower (3)	Higher (4)	Lower (5)	Higher (6)	Lower (7)	Higher (8)
Real interest rate	0.009***	-0.011***						
Nominal interest rate	(0.005)	(0.004)	0.008^{**}	-0.010** (0.004)				
Inflation expectation			-0.024*	0.030*	-0.024* (0.012)	0.027*	-0.033***	0.038^{***}
Score			(0.011)	(0.017)	0.016***	-0.018***	0.011***	-0.013***
Expected labor deman	d in next 3 mo	onths (omitted	category "ur	nchanged")	(0.001)	(0.001)	(0.002)	(0.005)
Lower	0.118***	-0.114***	0.118***	-0.114***	0.088***	-0.083***	0.103***	-0.096***
	(0.024)	(0.020)	(0.023)	(0.020)	(0.020)	(0.017)	(0.013)	(0.011)
Higher	-0.089***	0.146***	-0.088***	0.146***	-0.101***	0.149***	-0.090***	0.135***
	(0.017)	(0.030)	(0.017)	(0.030)	(0.015)	(0.025)	(0.011)	(0.019)
Expected total demand	l in next 3 mo	nths (omitted	category "un	changed")				
Lower	0.059*	-0.062**	0.059*	-0.063**	0.096***	-0.086***	0.082***	-0.080***
	(0.031)	(0.028)	(0.031)	(0.028)	(0.029)	(0.021)	(0.017)	(0.014)
Higher	-0.045***	0.061***	-0.044***	0.062***	-0.053***	0.066***	-0.041***	0.053***
	(0.015)	(0.023)	(0.015)	(0.023)	(0.014)	(0.019)	(0.009)	(0.013)
Expected price change	in next 12 m	onths						
	-0.003*	0.003**	-0.003*	0.003*	-0.002*	0.003*	-0.003***	0.004***
	(0.001)	(0.002)	(0.001)	(0.002)	(0.001)	(0.002)	(0.001)	(0.001)
Factors affecting exped	cted price cha	nge in next 12	2 months:					
Cost of labor	-0.004	0.005	-0.004	0.006	-0.017**	0.019**	-0.011**	0.013**
	(0.007)	(0.009)	(0.007)	(0.009)	(0.007)	(0.008)	(0.004)	(0.005)
Raw materials price	-0.005	0.006	-0.005	0.006	-0.004	0.005	0.003	-0.003
	(0.006)	(0.008)	(0.006)	(0.008)	(0.006)	(0.007)	(0.004)	(0.005)
Current conditions to i	nvest compare	ed with previo	ous 3 months	(omitted cate	egory "uncha	nged")		
Worse	0.113***	-0.110***	0.112***	-0.110***	0.118***	-0.105***	0.086***	-0.083***
	(0.030)	(0.023)	(0.029)	(0.023)	(0.027)	(0.020)	(0.016)	(0.014)
Better	-0.081***	0.131***	-0.081***	0.131***	-0.057***	0.074***	-0.080***	0.116***
	(0.018)	(0.034)	(0.018)	(0.033)	(0.019)	(0.027)	(0.012)	(0.020)
Current access condition	ons to credit c	ompared with	previous 3 n	nonths (omitt	ed category '	'unchanged'')	
Worse	0.065**	-0.070***	0.065**	-0.071***	0.059**	-0.058***	0.060***	-0.061***
	(0.026)	(0.024)	(0.026)	(0.025)	(0.023)	(0.020)	(0.015)	(0.014)
Better	-0.081***	0.129***	-0.081***	0.130***	-0.068***	0.091***	-0.065***	0.091***
	(0.016)	(0.030)	(0.016)	(0.030)	(0.017)	(0.025)	(0.013)	(0.021)
N. employees (log)	0.008	-0.011	0.008	-0.010	0.008*	-0.009*	-0.002	0.002
	(0.006)	(0.007)	(0.006)	(0.007)	(0.005)	(0.005)	(0.004)	(0.004)
Italy's current econom	ic outlook con	npared with p	previous 3 mo	nths (omittee	d category "u	nchanged")		
Worse	0.028	-0.032	0.028	-0.033	0.035	-0.036*	0.017	-0.019
	(0.023)	(0.026)	(0.023)	(0.026)	(0.021)	(0.021)	(0.013)	(0.014)
Better	-0.036*	0.050	-0.035*	0.049	-0.049***	0.062**	-0.022*	0.027
	(0.021)	(0.031)	(0.021)	(0.031)	(0.019)	(0.026)	(0.013)	(0.017)

Appendix Table 9.	Exploring the	Channels: Ro	ole of the	e Interest Rate
-------------------	---------------	--------------	------------	-----------------

Continues from previous page

Probability of an improve	ment in Italy	's general ecc	onomic situat	ion in next 3	months (om	itted category	"Zero")	
1-25 percent	0.004	-0.005	0.004	-0.005	0.008	-0.009	-0.013	0.016
	(0.015)	(0.020)	(0.015)	(0.020)	(0.014)	(0.016)	(0.009)	(0.011)
26-50 percent	0.002	-0.002	0.004	-0.005	0.022	-0.024	-0.013	0.015
	(0.024)	(0.031)	(0.024)	(0.031)	(0.023)	(0.025)	(0.016)	(0.019)
51-75 percent	0.068	-0.075*	0.070	-0.077*	0.032	-0.035	0.008	-0.009
	(0.043)	(0.041)	(0.043)	(0.041)	(0.033)	(0.034)	(0.023)	(0.025)
75-99 percent	0.078	-0.085	0.081	-0.087	-0.026	0.032	0.003	-0.003
	(0.093)	(0.085)	(0.094)	(0.085)	(0.068)	(0.090)	(0.062)	(0.069)
Share of revenues from ex	xports (omitte	ed category "() percent")					
0-33 percent	0.002	-0.003	0.003	-0.004	-0.020	0.023	-0.012	0.014
	(0.018)	(0.022)	(0.018)	(0.022)	(0.016)	(0.019)	(0.011)	(0.013)
33-66 percent	0.015	-0.019	0.017	-0.020	0.001	-0.001	0.011	-0.013
	(0.022)	(0.026)	(0.022)	(0.026)	(0.020)	(0.022)	(0.013)	(0.015)
66 percent and over	-0.011	0.014	-0.010	0.014	-0.029	0.035	-0.029**	0.037**
	(0.022)	(0.029)	(0.022)	(0.029)	(0.020)	(0.024)	(0.014)	(0.017)
Geographical area (omitte	ed category "	North-West")						
North-East	-0.031*	0.043**	-0.030*	0.041*	-0.020	0.024	-0.004	0.005
	(0.016)	(0.022)	(0.016)	(0.022)	(0.016)	(0.018)	(0.011)	(0.012)
Centre	-0.025	0.033	-0.023	0.031	-0.047***	0.060***	0.003	-0.004
	(0.018)	(0.025)	(0.018)	(0.025)	(0.017)	(0.021)	(0.011)	(0.013)
South and Islands	-0.041**	0.057**	-0.039**	0.055**	-0.031*	0.038*	-0.017	0.020
	(0.018)	(0.025)	(0.018)	(0.025)	(0.017)	(0.021)	(0.011)	(0.013)
Observations	3 137		3 137		3.080		8 9/6	
Pseudo R-square	0.043		0.044		0.047		0.082	

Notes. In this table we report the marginal effects on the predicted probability of reporting lower or higher investment expenditure of explanatory variables (expect for the sector and time fixed effects). In columns (1) and (2) the results are obtained using specification (2); in columns (3) and (4) the results are obtained using specification (3); in columns (5)-(8) the results are obtained using specification (4). In columns (5) and (6) the estimation sample is the same as in columns (1)-(4) whereas in columns (7) and (8) the estimation sample uses all available observations. The marginal effects are computed using the sample mean values for the continuous explanatory variables and the sample model categories for the qualitative explanatory variables. ***, **, * denote statistical significance at 1, 5 and 10 percent level. Robust standard errors are in parentheses.

	Ma	rginal effects	of 6-month in	nflation expe	ctations on p	redicted prob	ability to inv	rest
Regressors	Lower (1)	Higher (2)	Lower (3)	Higher (4)	Lower (5)	Higher (6)	Lower (7)	Higher (8)
Real interest rate	0.009***	-0.012***						
Nominal interest rate	(0.000)	(0.00 !)	0.008** (0.003)	-0.010** (0.004)				
Inflation expectation			-0.031* (0.016)	0.038**	-0.029* (0.016)	0.035* (0.019)	-0.038*** (0.009)	0.045*** (0.011)
Score			(0.020)	(0.019*** (0.005)	-0.023*** (0.005)	0.011*** (0.002)	-0.012*** (0.003)
Expected labor deman	d in next 3 mo	onths (omitted	category "un	changed")	(00000)	(00000)	(0000-)	(0.000)
Lower	0.118***	-0.114*** (0.020)	0.118***	-0.114***	0.107^{***}	-0.103*** (0.021)	0.102***	-0.096*** (0.011)
Higher	-0.089^{***}	0.146***	-0.089***	0.146***	-0.092^{***}	0.147***	-0.090***	0.135***
Expected total demand	1 in next 3 mot	nths (omitted	category "un	(0.050)	(0.017)	(0.050)	(0.011)	(0.01))
Lower	0.059* (0.031)	-0.062** (0.028)	0.058* (0.031)	-0.061** (0.028)	0.070** (0.032)	-0.071*** (0.027)	0.082*** (0.017)	-0.080*** (0.014)
Higher	-0.045*** (0.015)	0.061*** (0.023)	-0.045*** (0.015)	0.061*** (0.023)	-0.046*** (0.016)	0.062*** (0.023)	-0.041*** (0.009)	0.053*** (0.013)
Expected price change	in next 12 m	onths				()	()	(,
	-0.003* (0.001)	0.003** (0.002)	-0.003* (0.001)	0.003* (0.002)	-0.003** (0.001)	0.003** (0.002)	-0.003*** (0.001)	0.004*** (0.001)
Factors affecting expe	cted price cha	nge in next 12	2 months:					
Cost of labor	-0.004	0.005	-0.004	0.006	-0.006	0.007	-0.011**	0.012**
Raw materials price	(0.007) -0.005 (0.006)	(0.009) 0.007 (0.008)	(0.007) -0.005 (0.006)	(0.009) 0.006 (0.008)	(0.007) -0.005 (0.007)	(0.009) 0.006 (0.008)	(0.004) 0.002 (0.004)	(0.005) -0.003 (0.005)
Current conditions to i	nvest compar	ed with previo	us 3 months	(omitted cate	(0.007)	(0.000) nged")	(0.004)	(0.005)
Worse	0.113*** (0.030)	-0.110*** (0.023)	0.112***	-0.110*** (0.023)	0.115***	-0.109*** (0.023)	0.086^{***}	-0.083***
Better	-0.081***	(0.023) 0.130^{***} (0.033)	-0.081***	0.129***	-0.079***	0.122***	-0.080***	(0.014) 0.117*** (0.020)
Current access conditi	ons to credit c	compared with	nrevious 3 n	onths (omitt	ed category '	(0.022) 'unchanged'')	(0.020)
Worse	0.065**	-0.070***	0.066**	-0.071***	0.072***	-0.075***	0.060***	-0.062***
	(0.026)	(0.024)	(0.026)	(0.024)	(0.026)	(0.024)	(0.015)	(0.014)
Better	-0.081***	0.129***	-0.081***	0.130***	-0.083***	0.128***	-0.066***	0.091***
	(0.016)	(0.030)	(0.016)	(0.030)	(0.016)	(0.030)	(0.013)	(0.021)
N. employees (log)	0.008	-0.010	0.008	-0.010	0.007	-0.009	-0.002	0.003
Italy, a automatic aconom	(0.000)	(0.007)	(0.000)	(0.007)	(0.000) 1 aatagamu "…	(0.007)	(0.004)	(0.004)
Worse			0 0 2 8 0 0 10 0 10 0 10 0 10 0 10 0 10		0.010		0.017	_0.010
W0136	(0.027)	(0.032)	(0.028	(0.032)	(0.013)	(0.022)	(0.017)	(0.019)
Better	-0.036*	0.050	-0.036*	0.049	-0.038*	0.052*	-0.022	0.027
	(0.021)	(0.031)	(0.021)	(0.031)	(0.021)	(0.031)	(0.013)	(0.017)

Appendix	Table	10. E	xploring	the	Channels:	Role of	of the	Interest	Rate
r r · · · · · · ·			·········						

Probability of an impro	vement in Ita	ly's general e	conomic situa	ation in next	3 months (or	nitted catego	ry "Zero")	
1-25 percent	0.004	-0.005	0.004	-0.005	0.002	-0.002	-0.014	0.016
	(0.015)	(0.020)	(0.015)	(0.020)	(0.016)	(0.020)	(0.009)	(0.011)
26-50 percent	0.001	-0.002	0.002	-0.003	0.008	-0.010	-0.015	0.017
	(0.024)	(0.031)	(0.024)	(0.031)	(0.025)	(0.031)	(0.016)	(0.019)
51-75 percent	0.067	-0.074*	0.069	-0.076*	0.063	-0.068*	0.005	-0.005
	(0.043)	(0.041)	(0.043)	(0.041)	(0.042)	(0.041)	(0.023)	(0.025)
75-99 percent	0.078	-0.084	0.079	-0.085	0.074	-0.079	0.001	-0.001
-	(0.093)	(0.085)	(0.094)	(0.085)	(0.092)	(0.084)	(0.062)	(0.069)
Share of revenues from	exports (omi	tted category	"0 percent")					
0-33 percent	0.002	-0.002	0.003	-0.004	0.004	-0.004	-0.012	0.014
-	(0.018)	(0.022)	(0.018)	(0.022)	(0.018)	(0.022)	(0.011)	(0.013)
33-66 percent	0.015	-0.018	0.015	-0.018	0.010	-0.012	0.010	-0.011
	(0.022)	(0.026)	(0.022)	(0.026)	(0.022)	(0.026)	(0.013)	(0.015)
66 percent and over	-0.011	0.014	-0.011	0.015	-0.018	0.024	-0.029**	0.037**
	(0.022)	(0.029)	(0.022)	(0.029)	(0.022)	(0.029)	(0.014)	(0.017)
Geographical area (omi	tted category	"North-West	")					
North-East	-0.032**	0.043**	-0.032**	0.043**	-0.029*	0.038*	-0.005	0.006
	(0.016)	(0.022)	(0.016)	(0.022)	(0.017)	(0.022)	(0.010)	(0.012)
Centre	-0.025	0.034	-0.024	0.032	-0.024	0.032	0.004	-0.004
	(0.018)	(0.025)	(0.018)	(0.025)	(0.019)	(0.025)	(0.011)	(0.013)
South and Islands	-0.041**	0.057**	-0.040**	0.055**	-0.029	0.039	-0.016	0.020
	(0.018)	(0.025)	(0.018)	(0.025)	(0.018)	(0.024)	(0.011)	(0.013)
Observations	3137		3137		3080		8946	
Pseudo R-square	0.043		0.044		0.047		0.082	

Notes. In this table we report the marginal effects on the predicted probability of reporting lower or higher investment expenditure of explanatory variables (expect for the sector and time fixed effects). In columns (1) and (2) the results are obtained using specification (2); in columns (3) and (4) the results are obtained using specification (3); in columns (5)-(8) the results are obtained using specification (4). In columns (5) and (6) the estimation sample is the same as in columns (1)-(4) whereas in columns (7) and (8) the estimation sample uses all available observations. The marginal effects are computed using the sample mean values for the continuous explanatory variables and the sample model categories for the qualitative explanatory variables. ***, **, * denote statistical significance at 1, 5 and 10 percent level. Robust standard errors are in parentheses.

RECENTLY PUBLISHED "TEMI" (*)

- N. 1179 *Labor market and financial shocks: a time varying analysis*, by Francesco Corsello and Valerio Nispi Landi (June 2018).
- N. 1180 On the unintended effects of public transfers: evidence from EU funding to Southern Italy, by Ilaria De Angelis, Guido de Blasio and Lucia Rizzica (June 2018).
- N. 1181 Always look on the bright side? Central counterparties and interbank markets during the financial crisis, by Massimiliano Affinito and Matteo Piazza (July 2018).
- N. 1182 *Knocking on parents' doors: regulation and intergenerational mobility*, by Sauro Mocetti, Giacomo Roma and Enrico Rubolino (July 2018).
- N. 1183 Why do banks securitise their assets? Bank-level evidence from over one hundred countries in the pre-crisis period, by Fabio Panetta and Alberto Franco Pozzolo.
- N. 1184 Capital controls spillovers, by Valerio Nispi Landi (July 2018).
- N.1185 The macroeconomic effects of an open-ended asset purchase programme, by Lorenzo Burlon, Alessandro Notarpietro and Massimiliano Pisani (July 2018).
- N.1186 Fiscal buffers, private debt and recession: the good, the bad and the ugly, by Nicoletta Batini, Giovanni Melina and Stefania Villa (July 2018).
- N. 1187 Competition and the pass-through of unconventional monetary policy: evidence from TLTROs, by Matteo Benetton and Davide Fantino (September 2018).
- N. 1188 Raising aspirations and higher education: evidence from the UK's Widening Participation Policy, by Lucia Rizzica (September 2018).
- N. 1189 *Nearly exact Bayesian estimation of non-linear no-arbitrage term structure models*, by Marcello Pericoli and Marco Taboga (September 2018).
- N. 1190 *Granular Sources of the Italian business cycle*, by Nicolò Gnocato and Concetta Rondinelli (September 2018).
- N.1191 *Debt restructuring with multiple bank relationships*, by Angelo Baglioni, Luca Colombo and Paola Rossi (September 2018).
- N.1192 Exchange rate pass-through into euro area inflation. An estimated structural model, by Lorenzo Burlon, Alessandro Notarpietro and Massimiliano Pisani (September 2018).
- N. 1193 *The effect of grants on university drop-out rates: evidence on the Italian case*, by Francesca Modena, Enrico Rettore and Giulia Martina Tanzi (September 2018).
- N. 1194 *Potential output and microeconomic heterogeneity*, by Davide Fantino (November 2018).
- N. 1195 Immigrants, labor market dynamics and adjustment to shocks in the Euro Area, by Gaetano Basso, Francesco D'Amuri and Giovanni Peri (November 2018).
- N.1196 Sovereign debt maturity structure and its costs, by Flavia Corneli (November 2018).
- N. 1197 Fiscal policy in the US: a new measure of uncertainty and its recent development, by Alessio Anzuini and Luca Rossi (November 2018).
- N.1198 *Macroeconomics determinants of the correlation between stocks and bonds*, by Marcello Pericoli (November 2018).
- N. 1199 *Bank capital constraints, lending supply and economic activity*, by Antonio M. Conti, Andrea Nobili and Federico M. Signoretti (November 2018).
- N. 1200 *The effectiveness of capital controls*, by Valerio Nispi Landi and Alessandro Schiavone (November 2018).
- N. 1201 Contagion in the CoCos market? A case study of two stress events, by Pierluigi Bologna, Arianna Miglietta and Anatoli Segura (November 2018).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

- ALBANESE G., G. DE BLASIO and P. SESTITO, *My parents taught me. evidence on the family transmission of values,* Journal of Population Economics, v. 29, 2, pp. 571-592, **WP 955 (March 2014).**
- ANDINI M. and G. DE BLASIO, *Local development that money cannot buy: Italy's Contratti di Programma,* Journal of Economic Geography, v. 16, 2, pp. 365-393, WP 915 (June 2013).
- BARONE G. and S. MOCETTI, *Inequality and trust: new evidence from panel data*, Economic Inquiry, v. 54, pp. 794-809, **WP 973 (October 2014).**
- BELTRATTI A., B. BORTOLOTTI and M. CACCAVAIO, *Stock market efficiency in China: evidence from the split-share reform*, Quarterly Review of Economics and Finance, v. 60, pp. 125-137, **WP 969** (October 2014).
- BOLATTO S. and M. SBRACIA, *Deconstructing the gains from trade: selection of industries vs reallocation of workers*, Review of International Economics, v. 24, 2, pp. 344-363, **WP 1037 (November 2015).**
- BOLTON P., X. FREIXAS, L. GAMBACORTA and P. E. MISTRULLI, *Relationship and transaction lending in a crisis*, Review of Financial Studies, v. 29, 10, pp. 2643-2676, **WP 917 (July 2013).**
- BONACCORSI DI PATTI E. and E. SETTE, *Did the securitization market freeze affect bank lending during the financial crisis? Evidence from a credit register*, Journal of Financial Intermediation, v. 25, 1, pp. 54-76, **WP 848 (February 2012).**
- BORIN A. and M. MANCINI, Foreign direct investment and firm performance: an empirical analysis of *Italian firms*, Review of World Economics, v. 152, 4, pp. 705-732, WP 1011 (June 2015).
- BRAGOLI D., M. RIGON and F. ZANETTI, *Optimal inflation weights in the euro area*, International Journal of Central Banking, v. 12, 2, pp. 357-383, **WP 1045 (January 2016).**
- BRANDOLINI A. and E. VIVIANO, *Behind and beyond the (headcount) employment rate,* Journal of the Royal Statistical Society: Series A, v. 179, 3, pp. 657-681, WP 965 (July 2015).
- BRIPI F., *The role of regulation on entry: evidence from the Italian provinces*, World Bank Economic Review, v. 30, 2, pp. 383-411, WP 932 (September 2013).
- BRONZINI R. and P. PISELLI, *The impact of R&D subsidies on firm innovation*, Research Policy, v. 45, 2, pp. 442-457, **WP 960 (April 2014).**
- BURLON L. and M. VILALTA-BUFI, A new look at technical progress and early retirement, IZA Journal of Labor Policy, v. 5, WP 963 (June 2014).
- BUSETTI F. and M. CAIVANO, *The trend-cycle decomposition of output and the Phillips Curve: bayesian estimates for Italy and the Euro Area,* Empirical Economics, v. 50, 4, pp. 1565-1587, WP 941 (November 2013).
- CAIVANO M. and A. HARVEY, *Time-series models with an EGB2 conditional distribution*, Journal of Time Series Analysis, v. 35, 6, pp. 558-571, WP 947 (January 2014).
- CALZA A. and A. ZAGHINI, *Shoe-leather costs in the euro area and the foreign demand for euro banknotes,* International Journal of Central Banking, v. 12, 1, pp. 231-246, **WP 1039 (December 2015).**
- CESARONI T. and R. DE SANTIS, *Current account "core-periphery dualism" in the EMU*, The World Economy, v. 39, 10, pp. 1514-1538, **WP 996 (December 2014).**
- CIANI E., *Retirement, Pension eligibility and home production,* Labour Economics, v. 38, pp. 106-120, WP 1056 (March 2016).
- CIARLONE A. and V. MICELI, Escaping financial crises? Macro evidence from sovereign wealth funds' investment behaviour, Emerging Markets Review, v. 27, 2, pp. 169-196, WP 972 (October 2014).
- CORNELI F. and E. TARANTINO, *Sovereign debt and reserves with liquidity and productivity crises*, Journal of International Money and Finance, v. 65, pp. 166-194, **WP 1012 (June 2015).**
- D'AURIZIO L. and D. DEPALO, An evaluation of the policies on repayment of government's trade debt in *Italy*, Italian Economic Journal, v. 2, 2, pp. 167-196, **WP 1061 (April 2016).**
- DE BLASIO G., G. MAGIO and C. MENON, Down and out in Italian towns: measuring the impact of economic downturns on crime, Economics Letters, 146, pp. 99-102, WP 925 (July 2013).
- DOTTORI D. and M. MANNA, *Strategy and tactics in public debt management*, Journal of Policy Modeling, v. 38, 1, pp. 1-25, **WP 1005 (March 2015).**
- GERALI A., A. NOTARPIETRO and M. PISANI, *Macroeconomic effects of simultaneous implementation of reforms*, International Finance, Wiley Blackwell, v. 19, 1, pp. 42-65, **WP 997 (December 2014).**

- LIBERATI D., M. MARINUCCI and G. M. TANZI, Science and technology parks in Italy: main features and analysis of their effects on hosted firms, Journal of Technology Transfer, v. 41, 4, pp. 694-729, WP 983 (November 2014).
- MARCELLINO M., M. PORQUEDDU and F. VENDITTI, *Short-Term GDP forecasting with a mixed frequency dynamic factor model with stochastic volatility,* Journal of Business & Economic Statistics, v. 34, 1, pp. 118-127, **WP 896 (January 2013).**
- RODANO G., N. SERRANO-VELARDE and E. TARANTINO, *Bankruptcy law and bank financing*, Journal of Financial Economics, v. 120, 2, pp. 363-382, WP 1013 (June 2015).
- ZINNA G., Price pressures on UK real rates: an empirical investigation, Review of Finance, v. 20, 4, pp. 1587-1630, WP 968 (July 2014).

2017

- AABERGE, R., F. BOURGUIGNON, A. BRANDOLINI, F. FERREIRA, J. GORNICK, J. HILLS, M. JÄNTTI, S. JENKINS, J. MICKLEWRIGHT, E. MARLIER, B. NOLAN, T. PIKETTY, W. RADERMACHER, T. SMEEDING, N. STERN, J. STIGLITZ, H. SUTHERLAND, *Tony Atkinson and his legacy*, Review of Income and Wealth, v. 63, 3, pp. 411-444, WP 1138 (September 2017).
- ACCETTURO A., M. BUGAMELLI and A. LAMORGESE, *Law enforcement and political participation: Italy 1861-65*, Journal of Economic Behavior & Organization, v. 140, pp. 224-245, WP 1124 (July 2017).
- ADAMOPOULOU A. and G.M. TANZI, *Academic dropout and the great recession*, Journal of Human Capital, v. 11, 1, pp. 35–71, **WP 970 (October 2014).**
- ALBERTAZZI U., M. BOTTERO and G. SENE, *Information externalities in the credit market and the spell of credit rationing*, Journal of Financial Intermediation, v. 30, pp. 61–70, WP 980 (November 2014).
- ALESSANDRI P. and H. MUMTAZ, *Financial indicators and density forecasts for US output and inflation*, Review of Economic Dynamics, v. 24, pp. 66-78, **WP 977 (November 2014).**
- BARBIERI G., C. ROSSETTI and P. SESTITO, *Teacher motivation and student learning*, Politica economica/Journal of Economic Policy, v. 33, 1, pp. 59-72, WP 761 (June 2010).
- BENTIVOGLI C. and M. LITTERIO, Foreign ownership and performance: evidence from a panel of Italian firms, International Journal of the Economics of Business, v. 24, 3, pp. 251-273, WP 1085 (October 2016).
- BRONZINI R. and A. D'IGNAZIO, *Bank internationalisation and firm exports: evidence from matched firmbank data*, Review of International Economics, v. 25, 3, pp. 476-499 WP 1055 (March 2016).
- BRUCHE M. and A. SEGURA, *Debt maturity and the liquidity of secondary debt markets*, Journal of Financial Economics, v. 124, 3, pp. 599-613, WP 1049 (January 2016).
- BURLON L., *Public expenditure distribution, voting, and growth,* Journal of Public Economic Theory,, v. 19, 4, pp. 789–810, **WP 961 (April 2014).**
- BURLON L., A. GERALI, A. NOTARPIETRO and M. PISANI, Macroeconomic effectiveness of non-standard monetary policy and early exit. A model-based evaluation, International Finance, v. 20, 2, pp. 155-173, WP 1074 (July 2016).
- BUSETTI F., *Quantile aggregation of density forecasts*, Oxford Bulletin of Economics and Statistics, v. 79, 4, pp. 495-512, **WP 979 (November 2014).**
- CESARONI T. and S. IEZZI, *The predictive content of business survey indicators: evidence from SIGE,* Journal of Business Cycle Research, v.13, 1, pp 75–104, **WP 1031 (October 2015).**
- CONTI P., D. MARELLA and A. NERI, Statistical matching and uncertainty analysis in combining household income and expenditure data, Statistical Methods & Applications, v. 26, 3, pp 485–505, WP 1018 (July 2015).
- D'AMURI F., *Monitoring and disincentives in containing paid sick leave*, Labour Economics, v. 49, pp. 74-83, WP 787 (January 2011).
- D'AMURI F. and J. MARCUCCI, *The predictive power of google searches in forecasting unemployment,* International Journal of Forecasting, v. 33, 4, pp. 801-816, **WP 891 (November 2012).**
- DE BLASIO G. and S. POY, *The impact of local minimum wages on employment: evidence from Italy in the* 1950s, Journal of Regional Science, v. 57, 1, pp. 48-74, WP 953 (March 2014).
- DEL GIOVANE P., A. NOBILI and F. M. SIGNORETTI, Assessing the sources of credit supply tightening: was the sovereign debt crisis different from Lehman?, International Journal of Central Banking, v. 13, 2, pp. 197-234, WP 942 (November 2013).

- DEL PRETE S., M. PAGNINI, P. ROSSI and V. VACCA, Lending organization and credit supply during the 2008–2009 crisis, Economic Notes, v. 46, 2, pp. 207–236, WP 1108 (April 2017).
- DELLE MONACHE D. and I. PETRELLA, Adaptive models and heavy tails with an application to inflation forecasting, International Journal of Forecasting, v. 33, 2, pp. 482-501, WP 1052 (March 2016).
- FEDERICO S. and E. TOSTI, *Exporters and importers of services: firm-level evidence on Italy*, The World Economy, v. 40, 10, pp. 2078-2096, **WP 877 (September 2012).**
- GIACOMELLI S. and C. MENON, Does weak contract enforcement affect firm size? Evidence from the neighbour's court, Journal of Economic Geography, v. 17, 6, pp. 1251-1282, WP 898 (January 2013).
- LOBERTO M. and C. PERRICONE, *Does trend inflation make a difference?*, Economic Modelling, v. 61, pp. 351–375, **WP 1033 (October 2015).**
- MANCINI A.L., C. MONFARDINI and S. PASQUA, *Is a good example the best sermon? Children's imitation of parental reading*, Review of Economics of the Household, v. 15, 3, pp 965–993, **D No. 958** (April 2014).
- MEEKS R., B. NELSON and P. ALESSANDRI, *Shadow banks and macroeconomic instability*, Journal of Money, Credit and Banking, v. 49, 7, pp. 1483–1516, **WP 939 (November 2013).**
- MICUCCI G. and P. ROSSI, *Debt restructuring and the role of banks' organizational structure and lending technologies*, Journal of Financial Services Research, v. 51, 3, pp 339–361, **WP 763 (June 2010).**
- MOCETTI S., M. PAGNINI and E. SETTE, *Information technology and banking organization*, Journal of Journal of Financial Services Research, v. 51, pp. 313-338, WP 752 (March 2010).
- MOCETTI S. and E. VIVIANO, *Looking behind mortgage delinquencies*, Journal of Banking & Finance, v. 75, pp. 53-63, **WP 999 (January 2015).**
- NOBILI A. and F. ZOLLINO, A structural model for the housing and credit market in Italy, Journal of Housing Economics, v. 36, pp. 73-87, WP 887 (October 2012).
- PALAZZO F., Search costs and the severity of adverse selection, Research in Economics, v. 71, 1, pp. 171-197, WP 1073 (July 2016).
- PATACCHINI E. and E. RAINONE, Social ties and the demand for financial services, Journal of Financial Services Research, v. 52, 1–2, pp 35–88, WP 1115 (June 2017).
- PATACCHINI E., E. RAINONE and Y. ZENOU, *Heterogeneous peer effects in education*, Journal of Economic Behavior & Organization, v. 134, pp. 190–227, WP 1048 (January 2016).
- SBRANA G., A. SILVESTRINI and F. VENDITTI, *Short-term inflation forecasting: the M.E.T.A. approach,* International Journal of Forecasting, v. 33, 4, pp. 1065-1081, **WP 1016 (June 2015).**
- SEGURA A. and J. SUAREZ, *How excessive is banks' maturity transformation?*, Review of Financial Studies, v. 30, 10, pp. 3538–3580, WP 1065 (April 2016).
- VACCA V., An unexpected crisis? Looking at pricing effectiveness of heterogeneous banks, Economic Notes, v. 46, 2, pp. 171–206, WP 814 (July 2011).
- VERGARA CAFFARELI F., One-way flow networks with decreasing returns to linking, Dynamic Games and Applications, v. 7, 2, pp. 323-345, WP 734 (November 2009).
- ZAGHINI A., A Tale of fragmentation: corporate funding in the euro-area bond market, International Review of Financial Analysis, v. 49, pp. 59-68, WP 1104 (February 2017).

2018

- ADAMOPOULOU A. and E. KAYA, *Young Adults living with their parents and the influence of peers*, Oxford Bulletin of Economics and Statistics, v. 80, pp. 689-713, WP 1038 (November 2015).
- BARONE G., G. DE BLASIO and S. MOCETTI, *The real effects of credit crunch in the great recession: evidence from Italian provinces*, Regional Science and Urban Economics, v. 70, pp. 352-59, WP 1057 (March 2016).
- BELOTTI F. and G. ILARDI Consistent inference in fixed-effects stochastic frontier models, Journal of Econometrics, v. 202, 2, pp. 161-177, WP 1147 (October 2017).
- BERTON F., S. MOCETTI, A. PRESBITERO and M. RICHIARDI, *Banks, firms and jobs,* Review of Financial Studies, v. 31, 6, pp. 2113-2156, WP 1097 (February 2017).
- BOFONDI M., L. CARPINELLI and E. SETTE, *Credit supply during a sovereign debt crisis*, Journal of the European Economic Association, v.16, 3, pp. 696-729, **WP 909 (April 2013).**

- BOKAN N., A. GERALI, S. GOMES, P. JACQUINOT and M. PISANI, EAGLE-FLI: a macroeconomic model of banking and financial interdependence in the euro area, Economic Modelling, v. 69, C, pp. 249-280, WP 1064 (April 2016).
- BRILLI Y. and M. TONELLO, Does increasing compulsory education reduce or displace adolescent crime? New evidence from administrative and victimization data, CESifo Economic Studies, v. 64, 1, pp. 15–4, WP 1008 (April 2015).
- BUONO I. and S. FORMAI *The heterogeneous response of domestic sales and exports to bank credit shocks,* Journal of International Economics, v. 113, pp. 55-73, WP 1066 (March 2018).
- BURLON L., A. GERALI, A. NOTARPIETRO and M. PISANI, Non-standard monetary policy, asset prices and macroprudential policy in a monetary union, Journal of International Money and Finance, v. 88, pp. 25-53, WP 1089 (October 2016).
- CARTA F. and M. DE PHLIPPIS, You've Come a long way, baby. Husbands' commuting time and family labour supply, Regional Science and Urban Economics, v. 69, pp. 25-37, WP 1003 (March 2015).
- CARTA F. and L. RIZZICA, *Early kindergarten, maternal labor supply and children's outcomes: evidence from Italy*, Journal of Public Economics, v. 158, pp. 79-102, WP 1030 (October 2015).
- CASIRAGHI M., E. GAIOTTI, L. RODANO and A. SECCHI, A "Reverse Robin Hood"? The distributional implications of non-standard monetary policy for Italian households, Journal of International Money and Finance, v. 85, pp. 215-235, WP 1077 (July 2016).
- CECCHETTI S., F. NATOLI and L. SIGALOTTI, *Tail co-movement in inflation expectations as an indicator of anchoring*, International Journal of Central Banking, v. 14, 1, pp. 35-71, WP 1025 (July 2015).
- CIANI E. and C. DEIANA, *No Free lunch, buddy: housing transfers and informal care later in life*, Review of Economics of the Household, v.16, 4, pp. 971-1001, **WP 1117 (June 2017).**
- CIPRIANI M., A. GUARINO, G. GUAZZAROTTI, F. TAGLIATI and S. FISHER, *Informational contagion in the laboratory*, Review of Finance, v. 22, 3, pp. 877-904, WP 1063 (April 2016).
- DE BLASIO G, S. DE MITRI, S. D'IGNAZIO, P. FINALDI RUSSO and L. STOPPANI, *Public guarantees to SME borrowing. A RDD evaluation*, Journal of Banking & Finance, v. 96, pp. 73-86, WP 1111 (April 2017).
- GERALI A., A. LOCARNO, A. NOTARPIETRO and M. PISANI, *The sovereign crisis and Italy's potential output*, Journal of Policy Modeling, v. 40, 2, pp. 418-433, **WP 1010 (June 2015).**
- LINARELLO A., Direct and indirect effects of trade liberalization: evidence from Chile, Journal of Development Economics, v. 134, pp. 160-175, WP 994 (December 2014).
- NUCCI F. and M. RIGGI, *Labor force participation, wage rigidities, and inflation,* Journal of Macroeconomics, v. 55, 3 pp. 274-292, WP 1054 (March 2016).
- SEGURA A., Why did sponsor banks rescue their SIVs?, Review of Finance, v. 22, 2, pp. 661-697, WP 1100 (February 2017).

FORTHCOMING

- ACCETTURO A., W. DI GIACINTO, G. MICUCCI and M. PAGNINI, *Geography, productivity and trade: does* selection explain why some locations are more productive than others?, Journal of Regional Science, WP 910 (April 2013).
- ALBANESE G., G. DE BLASIO and P. SESTITO, *Trust, risk and time preferences: evidence from survey data,* International Review of Economics, **WP 911 (April 2013).**
- APRIGLIANO V., G. ARDIZZI and L. MONTEFORTE, Using the payment system data to forecast the economic activity, International Journal of Central Banking, WP 1098 (February 2017).
- BELOTTI F. and G. ILARDI, Consistent inference in fixed-effects stochastic frontier models, Journal of Econometrics, WP 1147 (October 2017).
- CIANI E. and P. FISHER, *Dif-in-dif estimators of multiplicative treatment effects*, Journal of Econometric Methods, **WP 985 (November 2014).**
- COVA P., P. PAGANO and M. PISANI, Domestic and international macroeconomic effects of the Eurosystem Expanded Asset Purchase Programme, IMF Economic Review, WP 1036 (October 2015).
- D'AMURI F., Monitoring and disincentives in containing paid sick leave, Labour Economics, WP 787 (January 2011).

- D'IGNAZIO and C. MENON, *The causal effect of credit guarantees for SMEs: evidence from Italy,* Scandinavian Journal of Economics, **WP 900 (February 2013).**
- ERCOLANI V. and J. VALLE E AZEVEDO, *How can the government spending multiplier be small at the zero lower bound?*, Macroeconomic Dynamics, **WP 1174 (April 2018).**
- FEDERICO S. and E. TOSTI, *Exporters and importers of services: firm-level evidence on Italy*, The World Economy, **WP 877 (September 2012).**
- GERALI A. and S. NERI, *Natural rates across the Atlantic,* Journal of Macroeconomics, WP 1140 (September 2017).
- GIACOMELLI S. and C. MENON, *Does weak contract enforcement affect firm size? Evidence from the neighbour's court,* Journal of Economic Geography, **WP 898 (January 2013).**
- GIORDANO C., M. MARINUCCI and A. SILVESTRINI, *The macro determinants of firms' and households' investment: evidence from Italy*, Economic Modelling, **WP 1167 (March 2018).**
- NATOLI F. and L. SIGALOTTI, *Tail co-movement in inflation expectations as an indicator of anchoring,* International Journal of Central Banking, WP 1025 (July 2015).
- RIGGI M., Capital destruction, jobless recoveries, and the discipline device role of unemployment, Macroeconomic Dynamics, WP 871 (July 2012).
- SEGURA A., Why did sponsor banks rescue their SIVs?, Review of Finance, WP 1100 (February 2017).